Hear AND Understand

What You Need to Know about Hearing Loss and Hearing Aids

Eric Frederick, AuD

Contents

Dedication

About the Author

Introduction

- Chapter 1: How Am I Harmed by Waiting Too Long for Hearing Aids?
- Chapter 2: How Can I Choose a Great Audiologist?

Chapter 3: What Can I Expect in My First Visits to the Audiologist?

Chapter 4: Do I Even Need Hearing Aids? Understanding Your Hearing Test

Chapter 5: Why Is Hearing Loss Harder to Correct Than Vision Problems?

Chapter 6: What Types of Help for My Hearing Loss Are Available?

Chapter 7: How Do Hearing Aids Work?

Chapter 8: What Costs and Legal Protections Will I Have When I Purchase Hearing Aids?

Chapter 9: Are Hearing Aids Really Worth All the Bother? Will They Really Help Me?

Dedication

To the thousands of patients who have entrusted me with improving their lives through better hearing. Your willingness to actively stay connected with your home, family, and community is a humbling inspiration every day.

About the Author

Dr. Eric Frederick is an audiologist with decades of work in non-profit hearing clinics, hospitals, ear, nose & throat surgery offices, and private practice audiology. He has worked in research and in clinical practice. He earned his Bachelor of Arts degree in Psychology from Indiana University in 1990; his Master of Science Degree in Audiology from Washington University in St. Louis in 1994; and his Doctor of Audiology degree from A.T. Still University of Health Sciences in 2002.

He has spent his time helping patients of all ages stay in touch with their world through better communication. He understands that hearing is not about just hearing speech, but is about keeping each and every person in touch with the people who are most important around them. There's a well-known quote by Hellen Keller stating that if she had to choose which sense to lose, she would rather be blind then deaf, because being blind cuts you off from a world of things, but being deaf cuts you off from a world of people.

Dr. Frederick has followed his passion for restoring hearing to keep people in touch with the most important people in their lives every day of his professional life. He currently is the founder and owner of Audiology Center Northwest in Portland, Oregon. His clinic is open to the public, and if you would like to contact Dr. Frederick or his team they are located at 919 Northeast 19th Avenue, Suite 170, Portland, Oregon 97232. His phone number is 503-232-1845. His Website is <u>www.audiologycenternw.com</u>, and emails may be directed to info@audiologycenternw.com.

Introduction

If you picked up this book, the chances are good that you or someone you care about is struggling to hear and understand speech. You are not alone. Hearing loss is one of the most common chronic health conditions in the United States, but it is an invisible disorder. When you have a hearing loss, it's not obvious—there's no wheelchair, no white cane. There's no external sign for people to know that you are struggling. In fact, even you may not know that you are struggling to hear and understand.

When you have a hearing loss, it progresses slowly most of the time. There's no pain to let you know that it's coming on. You just lose a little more clarity in your hearing every day until it builds up to a point where you finally have to admit that you have an issue—that it's not that everybody mumbles but that maybe your hearing actually isn't quite what it used to be.

We used to say that hearing loss was the third most common chronic health condition, right after arthritis and high blood pressure. That was based on the assumption that 10 percent of the population had a significant hearing loss. We know now that the number is much, much higher. We had come to that 10 percent figure by simply asking people if they had a hearing loss. Doctors don't diagnose any other condition in health care by simply asking people if they have the condition. Your physician would never diagnose high blood pressure just by asking you if you had it. In addition, hearing loss comes on so slowly that many people don't know or suspect that they have a problem.

Johns Hopkins University recently did a study in which they randomly tested a large population to see who truly had hearing loss. The researchers found that close to 20 percent of the population over the age of 12 has a hearing loss in at least one ear. By the time we're over the age of 50, that number is upwards of 25 percent. By the time we're 80, we have approximately an 80 percent chance of having a hearing loss.

The problem is that our society has not caught up with this situation. Our medical community is not actively screening people for hearing loss or talking about how important it is. We know that hearing loss can have a devastating impact on quality of life and relationships. We know that it can have a massive impact on mental health, including depression and anxiety. And as you read this book you'll find that there is a link between hearing loss and dementia, Alzheimer's disease, memory loss, and cognitive decline. Even though our medical system does not routinely test for hearing loss, you have taken an important first step by picking up this book. You've read this far, and I hope you will read more.

When you have a hearing loss, it starts out slowly. Most people don't even realize the loss, because almost no one is completely deaf. Even those with a significant hearing loss can hear; they just can't understand. The reason for that is that hearing loss does not affect every pitch equally. It's common to have good hearing for low-pitched, bass sounds but very impaired hearing for high-frequency, high-pitched sounds.

When you have a high-pitched hearing loss, you'll hear but not understand. The reason is that those deep bass sounds that most people can hear tend to be vowels, while the high-pitched, shrill, squeaky sounds that hearing loss takes away tend to be consonants. With most hearing losses, you can hear that someone is talking, but because you don't hear the high-pitched, brassy consonants like *s*, *f*, *t*, or *th*, you are constantly trying to put together a jigsaw puzzle with 30 percent of the pieces missing. That can be exhausting.

At the end of the day you may feel ground down. You may frequently misunderstand speech. You may make funny mistakes, and people may laugh at you because you misheard something. These are the symptoms that we would expect with a high-frequency hearing loss. The good news is that this can almost always be improved through technology.

One of the things that drives me crazy as an audiologist is hearing a physician say to a patient, "You don't need help. You have a hearing loss, but it's normal for your age" or "You're 80 years old, so you *should* have a hearing loss," or even "You're 60 or 70 years old, so you should expect to have a hearing loss. It's normal."

This is one of the most limiting things that you as a patient can ever hear. Just because you have a hearing loss that is common to people of your age group does not mean that we should ignore it. No matter what your age, hearing loss can cut you off from the world. It puts you at risk for mental health disorders, it damages your relationships, and it reduces your earning potential in the work force.

If a disorder is a problem for people who are 40 years old, it's also a problem for people who are 70. Society does not give up on you just because you have started collecting your pension. There is no other medical disorder for which a doctor would tell you, "You're just old... deal with it." No physician would ever tell you, "You're 80 years old and have cataracts. Just stop reading,

and deal with it." No, the doctor would step in and treat your cataracts to improve your vision. If a physician is telling you that hearing loss is normal for your age and that you should not do anything about it, it's time to see an expert in hearing loss instead of that physician.

Another thing that holds people back from getting hearing devices is the misperception that no one has success with them. Almost every day in my clinic I see new patients who know that they have a hearing loss and are not functioning well in their communication environment but say, "I have a friend who has hearing aids, but they just sit in the drawer. Why would I want to get something that doesn't work?" Others say, "I bought a set of hearing aids over the Internet, and I feel like throwing them away. Why would another pair be any better?"

My answer to these patients is that hearing aids actually have an extremely high satisfaction rate. The largest study done on hearing aid satisfaction, the Marketrak series of studies, shows that overall, hearing devices have an 85 percent satisfaction rate. That means that 85 percent of wearers are either satisfied or highly satisfied with their hearing devices. Technology continues to improve, and since the time that study was last done we're finding that some devices can get as high as a 96 percent satisfaction rating. That means that 96 out of every 100 people who try these devices rate themselves as satisfied.

Few products in the marketplace have this level of satisfaction—not cell phone service, not televisions, not cars, and not physical therapy. It's time for us to give up the myth that hearing devices don't work. Keep in mind, though, that not every single person is an appropriate candidate for hearing devices. Also remember that hearing aids need to be tailored to your individual hearing loss, not purchased in a store or over the Internet. It is my hope that by

reading this book, you'll learn the questions to ask when you visit a professional to see if and how you might benefit—and change your life—with hearing aids.

For example, one of the most common questions is whether hearing aids can help hearing when there is background noise present. When you have a hearing loss, the most difficult listening environments are almost always moderately noisy areas such as restaurants, group meetings, family gatherings, or car travel. When people ask me whether hearing aids can help in these situations, I assure them that the answer is yes. Technology has come a long way from what you may remember. If you remember hearing aids that did not help with background noise, or if you had hearing aids in the past that weren't much help, it's time to come in and discuss this issue. Because hearing in noisy environments is such an important problem, we'll discuss it at length in Chapter 6.

Another thing that keeps many people from pursuing hearing devices is the fact that they carry a cost—a cost in terms of both time and emotion, but also a significant financial cost. Hearing aids are expensive. There's no other way to say it. But if you know your rights, which we will talk about throughout this book—particularly in Chapter 8—you will come to understand how to minimize your financial risk.

One of the most important ways you can make sure you spend your money wisely is to understand that every hearing device dispensed in the United States comes with a money-back trial period. One of the primary purposes of this book is to help you, as a consumer, to understand what you can do to protect yourself and to get the most out of your hearing aid trial. You'll also learn about the warranty period, during which a hearing aid can be repaired at no cost, as well as loss and damage insurance that will cover you if your hearing aid becomes lost or catastrophically damaged. Getting a hearing device is an investment. Taking that step can be nerve-racking, but knowing your rights can put you back in control.

You may be reading this book now because you received it from your physician or your audiologist after a hearing test. You may have heard me speak on the subject of hearing loss and hearing devices.

You may have a feeling that your hearing is not what it once was. This is quite possible, because a huge number of risk factors can set you up for a hearing loss. Our bodies are connected systems. Our ears don't operate in isolation, and many health conditions that affect our entire bodies also affect our ears.

All of the following dramatically increase the likelihood that you will have a hearing loss: having cardiovascular disease, diabetes, or a family history of hearing loss; being a smoker; taking most medications that carry the risk of dizziness, hearing loss, or ringing in the ears; having erectile dysfunction and taking medication for it; or being exposed to either occupational or recreational noise, gunfire, or military service.

If you fit any of these descriptions—or if your doctor has told you that you might have a hearing loss—and you haven't had your hearing tested, it's time to pick up the phone and make a call. My hope for you is that in reading this book you will gain the knowledge to get back in the

driver's seat. Until you know something about hearing loss and hearing aids, you will not even know where to begin or what questions to ask. By learning more—by educating yourself and reading this book—you will be able to be an integral part of the process when it is time for you to go in and get your hearing tested. You will know what questions are most important to ask as a patient, and you will be able to feel confident in asking them. Hearing aids can be a lifechanging event. Getting hearing aids is a major life decision, and you need to be part of it.

It is likely that you have been thinking about your hearing for some time before choosing to pick up this book, wondering if your hearing loss is bad enough to address. One of the most surprising things you will read in this book may be the concept that you can actually be harmed by waiting too long to address your hearing loss. Back in the 1960s and 1970s, we used to say that you should wait as long as possible before getting hearing aids—that you should wait to seek help until your hearing is so bad that you cannot carry on a conversation anymore.

Current research suggests that this was terrible advice. There now appears to be a significant risk in waiting too long to correct hearing loss. In the United States, the average person waits seven years from the time they suspect a hearing loss to the time that they get their hearing tested. During that waiting period, what we're finding now is that the brain—the person's neurological system—is becoming less and less efficient at understanding the speech sounds that it's missing out on.

Your brain needs information to stay healthy, and information comes to it in the form of sensory input—in this case, sound. If you have a hearing loss, you are not feeding your brain the full diet of the sounds it needs to stay healthy. Research has not yet determined whether hearing loss directly causes cognitive decline, dementia, or Alzheimer's disease, but we do see a strong link between hearing loss and some mental health disorders and quality of life issues.

I hope this book will get people to get their hearing tested sooner, so that they can make an informed decision about whether they're going to pursue hearing devices. Hearing loss affects all aspects of life. Don't let it affect yours one day longer by putting off your hearing test.

I have spent more than 20 years helping people with hearing loss. I've worked with thousands of patients. Almost all of them have come in with some level of anxiety about the process, and I've found one thing to be true: most people want to know more. I've spent my adult life helping my patients learn more about why their hearing is affecting them the way it is, how technology can improve their hearing problem, and what the limits of technology might be.

A book can never be a substitute for face-to-face conversation with your clinical audiologist, but I hope this will be a starting point for you. It is my opportunity to tell you what I think are the most important issues that you need to know to put you back in control. We begin in Chapter 1, where I answer the question, How am I harmed by waiting too long for hearing aids? Once you've learned the benefits of hearing aids, Chapter 2 answers what is probably your most important question: How can I choose a great audiologist?

Once you've chosen a provider, you might be wondering what to expect at your first visits to the audiologist. Chapter 3 explains what will happen when you visit your provider. You'll most likely have a hearing test during that visit, and Chapter 4 will take you through that process step by step and tell you how to interpret the results.

The correction of hearing loss is complicated. Many people expect it to parallel the process of getting glasses and are surprised at the differences. Chapter 5 discusses why hearing loss is harder to correct than vision problems.

Chapter 6 covers the types of help that are available for hearing loss, and Chapter 7 explains how hearing aids work.

As I've noted, hearing aids are expensive. Chapter 8 helps you be in control and minimize your risks by discussing the costs and legal protections you have when you purchase hearing aids.

By the time you've absorbed all this information, you may be wondering, will hearing aids really help me? Chapter 9 reassures you that contacting an audiologist may be the best move you can make.

Treating hearing loss is an extremely individual process. You are a unique person, and to get the most out of your hearing treatment, you need to talk with a professional who can uncover your particular needs as you communicate in the world.

Take ownership of your hearing loss. You've taken the first step: you took the initiative to pick this book up and start reading. That means that you are in the small minority of people who are taking action to help themselves hear better.

I am honored that you took the time to begin reading this book. I know that you'll have a lot of other questions as you read, but I've tried to anticipate as many of them as I can so that I can answer them along the way.

If you have any questions or wish to discuss anything in this book in more detail and are in the Portland, Oregon, area please call my office. We would be happy to see you and consult with you about your particular needs. And if you are in another part of the country, I cannot encourage you strongly enough to visit an audiologist's office locally and discuss your hearing problems and needs. Remember, you are in control of your health care.

Chapter 1

How Am I Harmed by Waiting Too Long for Hearing Aids?

[]

Untreated hearing loss can have wide-ranging, serious effects on your life – effects that you may not even have thought of.

One of the major industries in my area is high tech, and a prominent microchip manufacturer in the area is a large employer. One of my patients who works there, who I'll call Stephen, told me that he wasn't able to function in work meetings because he wasn't able to hear in noisy environments where multiple people were talking or at the end of a conference table.

He was becoming concerned because he had become stagnant in his position. Even though he had a high level of skill, he had been passed over for new promotions. He had lost some clients and messed up a few projects because, he felt, of his inability to hear. Stephen's income had suffered, and now his marriage was strained, due both to the loss of his income and to communication difficulties with his wife.

We talked through his needs and made sure his wife, Jan, was in on the process. Jan was enthusiastic about getting him the help that he needed. Once we put hearing devices on him, his entire life changed. He was much more confident at work. He was able to function well in work meetings, on the phone, and in face-to-face conversation.

Stephen's been promoted. Jan says their marriage has been great because now he can hear her in restaurants. They can go out and do the things they used to enjoy doing together that bonded

them. Jan feels valued because he stepped up and took the action and responsibility to do what he could for his hearing. And they're doing things that they never thought would be important to them before, such as traveling to Europe—which they'd avoided because he wouldn't be able to hear the tour guide and thus wouldn't enjoy the trip.

Hearing loss touches every aspect of our lives in ways that we cannot even imagine until we correct it. Although I'm not a licensed psychologist, social worker, or counselor, I can say that I have had a huge number of people come in and tell me that I saved or improved their marriages.

You probably picked up this book because you're wondering if hearing aids can help you. You may feel you have a hearing loss, or your doctor may have recommended that you look into hearing aids.

Before you read on, you may want to begin by taking the quiz below to see if you might benefit from hearing aids.

Do you feel that you have difficulties hearing, or are you curious about possible symptoms of hearing loss? Then take this short evaluation. It's quick, painless, and easy, and it could be your first step to better hearing and living life to its fullest. (This checklist does not replace a professional hearing test, but the results might indicate whether or not it is worth seeking advice from an audiologist.)

- 1. Do people seem to mumble or speak in a softer voice than they used to?
- 2. Do you feel tired or irritable after a long conversation?
- 3. Do you sometimes miss key words in a sentence, or frequently need to ask people to repeat themselves?

- 4. When you are in a group or in a crowded restaurant, is it difficult for you to follow the conversation?
- 5. When you are together with other people, does background noise bother you?
- 6. Do you often need to turn up the volume on your TV or radio?
- 7. Do you find it difficult to hear the doorbell or the telephone ring?
- 8. Is it difficult to carry on a telephone conversation?
- 9. Do you find it challenging to pinpoint the location of an object, such as an alarm clock or telephone, from the noise it makes?
- 10. Has someone close to you mentioned that you might have a problem with your hearing?

If you have answered "yes" to two or more of the above questions, you will benefit from a hearing consultation.

Don't delay. If left untreated, hearing loss is associated with loss of income, relationship difficulties, depression, fatigue, possibly dementia, and much more. You are too young to be struggling with all that. Want to hear more? Call your audiologist today for a hearing checkup.

Why should I get hearing aids now instead of waiting a couple of years?

This book's cover claims it has "answers to all your questions." I suspect, though, that "How am I harmed by waiting too long for hearing aids" isn't one of those questions. Please read on. This topic is important to the success of your hearing aids.

It's almost always easier to get adjusted to hearing aids sooner rather than later. The main reason is that you hear with your brain, not your ears. You don't actually hear a sound until the signal gets sent from your ear up to your brain. Unfortunately, the cells in your brain require a steady diet of stimulation in order to stay healthy. If you go for years without stimulating the brain cells that receive signals from your ears (called neurons), these cells will actually degrade and atrophy, or waste away, over time. Your brain is almost like a muscle. If you don't use it and exercise it, it gets weaker over time. The connections between the neurons in your brain can become weaker. Cells in your brain can actually die off or get reassigned from one task, such as hearing, to another task.

We often see patients who have put off the process of getting hearing aids for years or decades. Most people, in fact, wait seven years before seeking help for a hearing loss. They have essentially been starving their brains of input.

If you do not address a hearing loss over time, it's not that the hearing loss will get more severe. You don't lose your ability to hear soft sounds. What happens is that your ability to understand speech becomes worse. Your ability to hear in difficult listening conditions—amid background noise, for example—also becomes worse.

One of the ways we've found out about the changes that occur over time is by studying people who have a hearing loss in both ears but who get a hearing aid in just one ear, usually for financial reasons. Their understanding of speech in the unaided ear tends to degrade faster than in the aided ear. Their ability to understand speech tends to get worse in the ear that is not used as much. We think that this is because the brain's ability to process signals from that ear gets worse over time. The brain cells that are responsible for understanding speech on the unaided side aren't being used and eventually weaken or even die off.

We do find that if we put a hearing aid on the unaided ear at a later date, the person will recover some neurologic function. The hearing aid, though, is never quite as efficient as it would have

been if the hearing device had been in that ear earlier. It also takes much longer for the person to regain the function in the ear that went a long time without a hearing aid.

It logically follows that when someone with a hearing loss doesn't have a hearing aid in either ear, this same type of nerve cell degeneration occurs for both ears. The problem is that it takes a long time to get the neurologic function back, and it may never come back fully. The rule of thumb is that for every year you go without hearing aids, it will take you approximately a month to get back to your best neurologic function.

I've heard that hearing loss may be related to dementia. Is there any truth to that?

A growing body of research from John Hopkins University demonstrates a definite relationship between hearing loss and cognitive decline such as dementia, memory loss, and Alzheimer's disease. It appears that approximately 30 percent of age-related cognitive decline is related to hearing loss. Please note that I've said there's a *relationship* between hearing loss and some kinds of cognitive decline. We do not have enough data to say that hearing loss *causes* dementia or other kinds of cognitive decline, but there certainly is a relationship.

The hearing loss tends to occur first, and the cognitive loss follows. If you have a mild hearing loss that is untreated, you are at twice the risk of having dementia or other cognitive decline. If you have, say, a moderate hearing loss that's untreated, you have three times the risk of dementia. And if you have a severe hearing loss that's untreated, you have five times the risk of dementia.

Another way to say this is that on average, every 10 decibels (dB) of hearing loss you have gives you a 20 percent increase in the possibility of having dementia or cognitive decline. Every 25 dB of untreated hearing loss you have adds seven years to your cognitive age. So if you have a 50 dB hearing loss that you haven't treated and you're an 80-year-old, you're creating a brain age of 94. This is a big deal. None of us wants to experience dementia, memory loss, or other cognitive issues any sooner than we have to.

I wish we could say with certainty that hearing aids will help delay cognitive decline, but we don't have the research to definitively say that yet. Studies are being done right now to determine whether treating a hearing loss with hearing aids can help delay the onset of dementia or memory loss. The most important research on this topic is a long-term, 25 year study. In this study, groups of people who thought their hearing was normal were compared with one group who thought they had a hearing loss, but DID wear hearing aids, and another group who thought they had hearing loss but DID NOT wear hearing aids.

These three groups were compared over 25 years of their lives. The authors reported that the groups who reported either normal hearing, or who reported using hearing aids to treat their hearing loss had the same level of mental/cognitive ability throughout the 25 year period. By comparison, *the group that reported that they had a hearing loss, but DID NOT wear hearing aids lost more mental/cognitive ability over time.*

Science and statistics are hard to interpret sometimes. We cannot always say that one thing causes the other. What we do know at this point is that hearing loss is a definite predictor of cognitive decline. We also know that hearing aids may possibly have the potential to help put off cognitive decline, although it's important to keep in mind that this has not yet been scientifically proven.

How can hearing aids make my life richer?

In 1999, The National Council on Aging did a wonderful study on the effects of untreated hearing loss. They gathered over 2300 hearing-impaired people and split them into two groups. In one group were people who had a hearing loss and opted to treat their hearing loss with hearing aids. In the other group were people who had hearing loss but opted not to treat their hearing loss with hearing aids.

The researchers then asked both groups of people—the hearing aid users and nonusers questions about more than a hundred quality-of-life indicators. They asked them about their social relationships, their work function, depression, anxiety, paranoia, self-esteem, overall satisfaction with life, social isolation, irritability, and fatigue. They found that on almost every one of these indicators, the group of people who opted to treat their hearing loss with hearing aids did better than the group who opted not to treat their hearing loss.

This really brings to light the fact that hearing aids aren't just about making things louder. They aren't even just about making you more able to understand speech. Hearing aids are about making your life better—helping you to be the person that you see yourself as, the person you want to be.

The people in the study felt that because of their hearing aids, their relationships were better. They felt that their hearing aids gave them higher self-esteem or higher quality of life.

If you can communicate better, you're more likely to be more social. You're more likely to feel more confident. You're more likely to perform better at work. If you perform better at work,

you're more likely to make more money. You're more likely to be active in your community or engaged with your family. Hearing aids are about changing how you walk through the world.

A small percentage of people in the study even said that their love life got better after getting hearing aids. This little fact really shows just how far-reaching the ability to hear can be in our lives. After all, hearing aids are about connecting with other people, letting you get in touch with people. And if you can connect and communicate and stay in touch with your family more, you're more likely to be a better partner to your spouse. It's not surprising that your romantic relationships will get better if you're more responsive to your partner. After all, the best things we say to each other in life are said in a whisper, not in a shout.

How can I be more in charge of my hearing health?

It's your job, as the director of your own health care, to understand that many health conditions can cause hearing loss. In fact, your physician may not know about some of them. It's up to you to take a proactive stand and bring this topic up with your doctor.

Risk Factors for Hearing Loss

You may be at greater risk for hearing loss if any of the following apply to you:

- You have diabetes.
- You have cardiovascular disease. [list the common conditions that fall in this category]
- You have kidney disease
- You take certain medications, such as Viagra. [list others]
- You have a history of smoking.

• You have a history of alcohol abuse.

Hearing loss may put you at risk for the following:

- Decreased cognitive function
- Depression
- Falling
- Decreased earnings at work

Diabetes doubles your chance of hearing loss. Prediabetes also doubles your chance of hearing loss, as does kidney disease. Many cardiovascular diseases, or heart disease, triple your chance of hearing loss. Smoking also increases your chance of hearing loss, and hearing loss frequently even causes depression.

Alcohol is linked with hearing loss, as are many medications. For example, Viagra doubles your chance of hearing loss, and two doses a week of Tylenol or other over-the-counter pain relievers can also double your chance of hearing loss.

If you have chronic pain, kidney disease, heart disease, diabetes, or any of these other risk factors, it's up to you to go to your doctor and say that you want regular hearing testing. There's no reason for hearing loss to rob you of years of connection with the people who are most important to you.

Your physician may not realize all of these links between health conditions and hearing loss. Primary care doctors have a lot on their plate and don't always have hearing loss on their radar screen. You need to feel empowered; go in and tell your physician that you read this book and understand that there's a link between some health conditions and hearing loss. Ask that your doctor refer you for hearing testing on a regular basis. Take charge of your hearing health!

Chapter 2

How Can I Choose a Great Audiologist?

In choosing a hearing aid, probably the most important thing that you can do as a patient is to choose the right person to get your hearing aid from.

One of my patients, who I'll call Tammy, was a perfect case in point. I met her when I was doing some public speaking; she came up to me and told me about her last experience. She was in her early 50s and had purchased hearing aids about a year before. But she had never been happy with them. She'd been back to her provider several times to try make the hearing aids right, but she always felt that she heard better without them, and she was never really at ease with her provider.

Tammy came to me at the end of the talk and asked, "Eric, you seem like you know what you're doing. Would you be willing to take a look at my hearing aids and see if they can be made better for me?"

I said, "I'd be happy to do it. I'll start with a hearing test, and then we'll figure out what to do from there." The first time I saw her for the hearing test, I found out why she was unhappy. It turned out her hearing was normal.

She had been convinced by a hearing aid dealer to spend thousands of dollars on devices she didn't need. This, unfortunately, is one of the horror stories that we occasionally hear, and it's one of the reasons why hearing aids get such a bad reputation. There are different types of

professionals who fit hearing aids, and Tammy went to somebody who was not a medical professional. The hearing aid dealer she saw was a commissioned salesperson, whose income depended solely on how much money customers spent.

The hearing aid dealer talked her into buying something she didn't need and then kept convincing her that if she just came in for one more adjustment, everything would be perfect, until she was beyond the point where she could return the hearing aids for a refund. By the time I saw her, she intuitively knew what the situation was. She knew that if she was hearing better without the hearing aids than with them, she never needed them in the first place.

In the end, I had to deliver the bad news that there was no way to make the hearing aids better for her. Instead, she needed to hold onto them until she actually had a hearing loss, and then we could make the hearing aids work for her. She was disappointed that she had spent the money but was relieved that she didn't have the same experience with me—that I didn't tell her everything would be okay if she just spent some more money. In retrospect, Tammy realized that she had never been comfortable with the first provider. She'd always felt pressured to purchase something rather than to understand her hearing loss. If she had understood how to interpret her test results, which we'll discuss in Chapter 4, she never would have been in this position.

Selecting a professional. Choosing a professional to get your advice from is probably the most important single choice you have in getting treatment for your hearing aid. In the following pages we will explore how to stack the deck in your favor when deciding who should advise you on your hearing loss.

I think I may have a hearing loss. What should I do first?

If you suspect you have a hearing loss, you may be feeling overwhelmed with the number of choices. You may know various people with different kinds of hearing aids, and you may have noticed that there are different types of providers. What are the differences among them, and how do you choose? This chapter will give you the information and guidance you'll need.

It's important to keep in mind that when you buy hearing aids, you are a consumer, and you can be an active part of the process. By reading this book, you will find out how to be an informed consumer and learn what kinds of questions to ask that will get you to a successful outcome.

What kinds of specialists treat people with hearing loss?

When you start the process of selecting a provider, it's often hard to know exactly what type of professional to see. There are really different types of professionals who deal with hearing loss: an ear, nose, and throat physician, an audiologist, and a hearing instrument specialist (or hearing aid dealer).

When should I visit an ear, nose, and throat physician?

The most visible and well known of the professionals who treat people with hearing loss is an *ear, nose, and throat physician,* or ENT, also sometimes called otolaryngologists. If they specialize just in ears, they are called otologists. If they specialize in the ear and the brain, they are called neurotologists. All of these professionals are medical doctors. They've been to medical school and have been trained in treating hearing loss with medications and surgery. If you have a type of hearing loss that will respond well to medication or surgery, an ear, nose, and throat physician should absolutely be involved and should be the team leader in your treatment.

You should see an ENT as soon as possible if your hearing loss came on suddenly; for example, if you woke up one day and your hearing was much worse than the day before. This kind of hearing loss may respond very well to medication, but only if it's given in the early stages.

If you have a large difference between the hearing in your two ears, an ENT physician should be involved to determine why there's a difference between the two ears. The physician may use an imaging study of the head or neck, such as an MRI or a CT scan, to understand whether an anatomical difference explains the asymmetry of hearing.

In addition, if you're having pain in the ear, liquid draining from your ear, bleeding from your ear, significant dizziness, or a feeling of pressure in the ear, an ENT should be immediately involved in your care.

When should I visit an audiologist?

The reality, however, is that according to the Better Hearing Institute in Washington, D.C., only about 10 percent of hearing losses at any given time respond well to medication or surgery. The other 90 percent are permanent. The vast majority of people have a long-standing, gradually progressing hearing loss, and these people are best served by seeing an audiologist first.

Audiologists are highly trained professionals who have had anywhere from six to eight years of college to earn the title of audiologist. All audiologists coming into the field today need a doctorate degree. They typically have an AuD degree, which stands for doctor of audiology. You may find an older audiologist with a PhD or a master's degree.

Audiologists usually receive four years of training through a university after four years of undergraduate work in speech and hearing sciences. Their training encompasses hearing loss, anatomy and physiology, rehabilitation techniques, and hearing devices. In addition, audiologists are trained to help identify many medical conditions that are better treated by an ear, nose, and throat physician. If you start with an audiologist to assess possible hearing loss, they have the training to guide you to an ENT if you would be better served by such a specialist. You can feel confident that audiologists have a strong basis of education to do their job.

It's also fine if you decide to begin by going to an ENT. Almost every ear, nose, and throat physician has an audiologist on staff or one that they refer their patients to. To treat hearing loss, ENTs need to be able to diagnose it. They do so through a hearing test, which is typically performed by an audiologist.

As an audiologist, I never discourage patients from getting a medical second opinion from an ENT. The two professions work very well hand in hand. ENTs cover the medical side, and audiologists cover the rehabilitation side, so each professional knows something that the other one does not. As a starting point, though, most people will be served very well by beginning with a hearing test and then being referred to a primary care physician or an ear, nose, and throat physician if the audiologist finds that medical treatment might be helpful.

Should I ever choose to visit a hearing aid specialist or hearing aid dealer?

The third category of professionals who work with hearing loss and hearing aids are called *hearing instrument specialists, hearing aid dispensers, hearing aid dealers,* or even *audioprosthetologists*. In essence, they are all the same thing. The state you live in will

determine this professional's title.

This category of professional is much more loosely regulated than ear, nose, and throat physicians and audiologists. Hearing instrument specialists are generally governed by state law, and states vary greatly in the amount of training that they require. For example, I practice in Oregon, only about 10 minutes away from the state of Washington. An Oregon-based hearing instrument specialist by law must have two years more training than a Washington-based one, just one state over.

There are many very good hearing instrument specialists. However, keep in mind that these providers will have a much more variable course of training than you'll find with either an audiologist or an ear, nose, and throat physician. Most of the excellent hearing instrument specialists that I have encountered are fantastic at their jobs because they've gone above and beyond what the state requires for training.

How do I choose between an audiologist and a hearing instrument specialist to help with my hearing loss?

If you live in an area with access to both audiologists and hearing instrument specialists, why not stack the cards in your favor by going to somebody with more extensive education? An audiologist will virtually always have a much higher level of education than a hearing aid dealer.

If you do not live in an area with access to audiologists, be sure to find the most qualified hearing instrument specialist you can. One way to do so is to look for the letters *BC-HIS* after the person's name. *BC* stands for board certified, and *HIS* stands for hearing instrument

specialist. These providers have voluntarily gotten additional training and undergone a nationwide certification program for hearing instrument specialists.

In some states, including my state of Oregon, ear, nose, and throat physicians and audiologists are categorized as medical professionals, while hearing instrument specialists are categorized as retail salespeople. Because of the differences in training and business models, you may have a very different experience with an ENT or audiologist than with a hearing instrument specialist.

If you're the type of person who thrives in the health-care system, you might choose to begin with an ENT or an audiologist. This might suit you if you see hearing loss as a medical condition that needs to be treated as such.

If instead you thrive in a retail model, you may prefer to feel like a customer rather than a patient. You may want to go where you can negotiate the price of a hearing aid, which audiologists cannot do—in fact, no provider who bills insurance companies can negotiate prices. If a provider who accepts insurance coverage lowers his cost on a device through haggling, this is considered insurance fraud with possible fines and jail time as a punishment. You as the patient should be extremely wary if your hearing aid provider starts offering discounts when you express reluctance to make a purchase decision.

Should I buy hearing aids over the Internet?

Recently, more and more companies sell hearing aids over the Internet. This means they sell to prospective patients they've never seen and recommend hearing devices for patients they've never tested. Some do have the patient send in a hearing test performed by an audiologist;

someone looks at those test results and recommends a device that might be reasonably appropriate.

The problem here is that buying a hearing aid is not like buying a toaster or other appliance. Getting a hearing device is much more than just getting a product. The product itself—the hearing aid—accounts for only a small part of your success in using it. A hearing device is basically a blank slate. It must be programmed by a professional, who tells it how to behave when sound is coming in. The programming allows the hearing device to process sound and shape that sound in such a way that it gives you your best speech understanding while keeping things as comfortable as possible. This programming is as unique to you as your eyeglass prescription and will not sound right to someone else.

Ideally, we want soft speech to be audible, moderate speech to be comfortable, and loud speech to be tolerable. At the same time we want to make speech as clear as possible while keeping background noise to a minimum. This is a very complicated process; it's not something that you can do well right out of the box. The process of getting a hearing device is very much like physical therapy. Your ability to process sound will change over time as you use the hearing device. It's like building muscle. You'll be able to make use of sounds after a month of hearing aid wear that you could not make use of on the first day.

There is a dance that happens between the hearing device and the patient. As the patient gets more capable, the device should deliver more to meet that capability. As part of the hearing aid fitting process, you are going to need to go back to the dispensing professional a few times. Three to four times is pretty typical in the first couple of months, because your brain itself will

change as it gets more experience with the sounds that it hasn't experienced in years. You'll become a more efficient listener over time. And you'll want to feed your more efficient brain a higher quality signal so that you understand still more speech.

You can't get these adjustments if you purchase hearing aids over the Internet. You need to work with a professional you can visit for adjustments.

I see much lower prices for hearing aids in magazines and newspapers. Should I look into them?

It may be tempting to try to choose a hearing aid provider based on advertisements you see in newspapers or magazines. Beware: Buying hearing aids because their provider offers a low price can be one of the riskiest ways to pick a health-care provider.

Businesses pay many thousands of dollars to run flashy ads. Who pays for these ads? It's you, the patient, who pays. When you buy hearing aids from providers who buy these ads, you are paying their advertising budget.

In addition, the highest quality, best educated providers don't have to advertise aggressively in most communities. They get more than enough business through physician referrals and word of mouth from satisfied customers. These providers have treated patients well, so their patients recommend them to their friends. If a provider has to rely on large, flashy advertisements, you have to ask yourself why the business is not already full.

You may see what seem like amazing prices in advertisements. You may see two-for-one specials or \$1000-off deals. Think about it, though: if the hearing aids are a thousand dollars off or are half priced in a two-for-one special, the business may well have to overcharge people regularly to be able to bring the prices down that much.

If you call around and ask about prices, be very wary of any place that has much higher or much lower prices than everyone else. If you call five places and one of them is twice as expensive as everyone else, cross that provider off the list. Similarly, if you call five different places and one quotes you prices that are half of what everyone else charges, you need to be wary. Providers that quote or advertise extremely low prices probably have some hidden costs. When you go in, you may find that a lot of things are not included in the price you were quoted. For example, the impression of your ear to fit your hearing aids might not be included. Or the price might include only one or two visits, when actually you'll need several more. Or your warranty may be very short. Or you may not have as long a time to return the hearing aids for a refund. Or, if you do return them for a refund, there may be extremely high restocking fees for which you will be responsible.

In my local area here in Portland, if you look at the same hearing device from all the reputable sources, the price does not vary more than 5 or 10 percent. The reason is that most reputable providers charge what it actually costs to dispense a hearing device. They don't pad the bill, and they are very up front about costs.

In the hearing aid world, as in the world in general, you'll tend to get what you pay for. And you'll find that the most reputable people are all charging what it really costs to get you the

hearing aids that work best for you.

Some of my friends are not happy with their hearing aids. Is anyone really going to be able to fix my hearing anyway?

As an audiologist, I hear this over and over: "My friends say their hearing aids don't help much. Are hearing aids really something I want to pursue?"

The average patient coming in for a hearing test has waited seven years before getting tested. Most patients have known they have problems with hearing but have put off doing anything about it until they just can't stand it anymore. They may feel they are less happy with their social life or their professional life because of their hearing loss.

Many people look for excuses not to do something about their hearing problems. One of the biggest excuses they find is that they know people who have had bad experiences with hearing aids. They may have friends who keep their hearing aids in a drawer because they do not like them or know people who complain about not hearing well with their devices.

But remember that most people who talk about their experiences are people who are unhappy with them. Think back over the last month. How often did you tell somebody how wonderful it was that your feet, ankles, and legs work? That you could walk across the room without any pain—that it was a miracle of biology that you could get from here to there on your feet? We never tell anybody about our body's successes. But if we sprain an ankle, we tell everyone who listens how much pain we feel.

The same thing happens with hearing aids. The people who are happy with their experiences are generally the silent majority. The people who are unhappy, or who haven't been dealt with fairly by their providers, will tell everyone. And honestly, they *should* tell people. That's one of the ways people like you can figure out whether a prospective provider is a good or a bad risk. If you know five people who have had a negative experience with the same business, you'll know not to go there.

Some people have horror stories because they encountered a bad provider or malpractice, or because their hearing loss is not the kind that can be helped with hearing aids. Some kinds of hearing loss are not correctable with hearing devices. However, a good provider should be able to identify whether a person is a good candidate for hearing aids and will not dispense them if they will not help the person hear better. On the whole the vast majority of people are happy with their hearing aid provider. The horror stories we have spoken about really are the exception, not the rule.

Remember, if your friends are not satisfied with their hearing aids, don't think, "There's no point in getting hearing aids." Instead think, "I won't get my hearing assessed by that provider."

Hearing aids on the whole have an extremely high success rate. Approximately 81 percent of people who own hearing aids report being either satisfied or very satisfied. This rate compares favorably with almost any other consumer product. In fact, hearing aids have a higher satisfaction rate than cars, TVs, cell phone service, gourmet coffees, physical therapy, and Internet providers, and almost as high a satisfaction rate as Lasik surgery for the eyes, which is considered to be a near-perfect medical treatment.
In fact, certain hearing devices have up to a 96 percent satisfaction rate. That is absolutely phenomenal. There is almost nothing in this world that produces a 96 percent satisfaction rate. And there is certainly nothing else I know of that creates a 96 percent satisfaction rate but still has the widespread public perception that it doesn't work. If you have hearing loss, you owe it to yourself not to believe what everyone tells you but to go in and listen for yourself.

How can I choose the right provider and not get ripped off?

In the end, you need to pick a provider and a hearing device that works for you, and the only person who can determine that is you.

You'd do well to begin by asking other people what their experience has been with their providers. Ask your friends whom they use, and ask your physicians whom they would recommend.

When you begin to select among recommended providers, the most important thing to ask them is, "What options will I have to experience the device first?" You should be able to try a hearing device in the office before you even decide whether to order one. After all, you test drive a car before you buy it; why shouldn't you be able to test drive your hearing aids? In addition, the provider should give you a period of time when you can try the hearing device and return it for a refund. A competent provider will give you different ways that you, as the patient sitting in the chair, can understand and experience what you're getting into before you spend your money.

When I advise people how to pick a provider, I give some rules of thumb that are almost exactly

First, if possible, go to an audiologist for your hearing tests. Audiologists have the highest level of training in hearing loss and how to treat it with devices.

Second, make sure that you get your hearing tested in a sound-treated booth or room. The office must have a dedicated space to run the hearing test in a very quiet environment. There must be a sound enclosure to isolate you and test your hearing in a quiet environment, not in a noisy office.

Third, go to an audiologist who works with multiple brands of hearing aids instead of just one or two. Some hearing aid providers have a financial arrangement with a certain manufacturer they are either owned or funded by the manufacturer, and they can only order one brand of product. That's not always the best thing for a patient. Each manufacturer has its own philosophy and strategy for how to deal with hearing loss. Being able to select from a wide variety of products means that a provider can handle a wide variety of patients with many kinds of hearing losses. The provider should choose what's right for you, not force you to take what a particular manufacturer produces. You are an individual, and your provider should treat you as such.

Fourth, the audiologist should make sure that you are getting the most benefit from your hearing devices. At the least, they should ensure that you are actually performing better with the hearing aids on than with them off. Too many providers simply take what the manufacturer recommends for your hearing loss, put the devices on your ears, and tell you to get used to them. This is generally what one would experience with Internet sales as well. That should never happen. When you are getting hearing health care, you are the most important person in the room. Your

provider should sit down with you and make sure that the sound that is being delivered to you is making you communicate more effectively and keeping you comfortable.

Fifth, the audiologist should always take their time with you. They should talk with you in-depth about how to use the hearing aids, how to adapt to them, and how to maintain them. Although it's not a terribly difficult process, learning to use and maintain hearing aids does take some practice. It doesn't feel natural on day one. You will no doubt have some questions during the first month or two. If you ever get the idea that your provider doesn't have time for you after you've written the check, that might not be the right person to continue working with.

Every hearing aid sold in the United States has at least a 30-day trial period. If you're not happy with your provider during your trial period, return the hearing aids and find a different provider. If your trial period is over, it's perfectly acceptable to find a different audiologist to take over your care. Chapter 8 discusses your legal protections in buying hearing aids in more detail.

You are in control of your hearing health. Don't settle for a provider who you feel is not allowing you to have the best hearing possible for you.

Chapter 3

What Can I Expect in My First Visits to the Audiologist?

Once you've found an audiologist that you like, I'm sure you'll have a number of questions about the process. Every audiologist office operates just a little bit differently.

Now that I've found an audiologist I'd like to try, what should I do next?

You'll want to call the office and talk to the staff in the front office. Have your questions ready. You may want to ask how many appointments you should expect in the first few weeks and how much time each appointment might take. You may want to ask about basic price ranges to get an idea of a possible budget. And you may want to discuss possible insurance coverage. We'll cover all these topics in this book.

If the office is reluctant to talk about cost, it should be a warning sign. If you call an office and ask, "How much can I reasonably expect to spend on hearing aids, at a minimum, if I come in?" the office should be willing to give you at least the low and high ends of the price range. When a patient calls my office, we make sure to discuss what low-priced, mid-priced, and high-priced hearing devices would cost, and what the patient could reasonably expect to get for the additional money. That way, when patients come in they have a good idea of what to budget and why they might want to climb up in price a little bit.

How much time should I expect to set aside for my appointments?

In general, a hearing test takes about half an hour. That includes some questions about your medical history, the test itself, and talking over the test results. If you already think you have a hearing loss and want to discuss hearing aids, most offices will budget about an hour for your first visit. That will give you time to do the hearing test and talk about your options in hearing devices.

If you feel rushed to make a decision at your first appointment, or if you feel you need more information, by all means make a follow-up appointment. You should never feel pressured to make a decision on day one. You may need time to think about it and talk it over with your family. Don't hesitate to let your audiologist know that you need time and would like to come back and ask follow-up questions on a later day.

Be sure to ask how many appointments you are likely to need in the first couple of months. Getting hearing devices is not an event; it is a process. It's something that takes more than one visit, so you'll want to make sure that the audiologist is ready to spend the time that you need to make adjustments to the device and to ensure that you're hearing sound that feels the most comfortable and the most clear to you.

After the initial hearing test, you will typically need from two to five visits in the next couple of months. These visits are designed to customize the sound of the hearing devices to meet your needs. Be prepared to put in the time to make this experience what you want it to be. You need to feel comfortable talking to the audiologist and reporting any abrasive sounds or sounds that you're not hearing that you feel you should.

Getting hearing aids is a tailor-made process for each and every patient. Remember, though, that your audiologist can't help you unless you let them know what's going on.

Will I know ahead of time what my insurance will cover?

In the past, it was very uncommon for health insurance to cover hearing aids. Today, though, we're definitely seeing a change in the willingness of health insurance companies to cover at least a portion of the cost of hearing aids.

Your audiologist's staff should be able to help you navigate the insurance system. They should be able to tell you, in very clear language, whether your insurance will cover any of the costs. All they will need is the information from your health insurance card to check it out. It's reasonable to ask the office to determine your insurance coverage before your first appointment.

Insurance coverage and costs involved with hearing aids are covered in Chapter 8.

What should I bring along on my first visit to the audiologist?

I would recommend that when you come in for your first visit—the hearing evaluation—you bring a spouse or family member with you. You may be making an expensive life decision, and it always helps to have someone else's input. If that someone else is affected by your hearing loss as well, it's even better to include that person. Communication is a two-way street, and typically the person who is most affected by the hearing loss is a spouse or partner. If you and your partner go in and talk to an audiologist, you may find that your partner noticed your hearing loss long before you did and feels that it is creating more problems in your life than you are aware of. Having that person there with you often will give you the license to go ahead and make a decision to do the right thing for yourself.

You'll also want to bring along your medical records or at least notes on your medical history. At your first visit, your audiologist will probably ask you to fill out a form describing your medical history. This is very important. Most people don't understand why we ask questions about disorders that are seemingly unrelated to hearing loss. We often have people ask, "Why do you want to know if I have high blood pressure, heart disease, or diabetes, or what medications I'm on?" The simple answer is that these things often are related to hearing loss.

As we discussed in Chapter 1, many disorders and medications that we would not think would have an effect on us actually affect our entire body system, including our ears. Diabetes doubles your chance of hearing loss, and cardiovascular disease triples your chance. Many medications that we wouldn't think of, including Tylenol and Viagra, increase your chances of hearing loss. It's important for your audiologist to know what medications you take so that when discussing either medical referrals or device options, the audiologist has some idea of how rapidly your hearing loss is likely to progress. Additionally, the audiologist may be able to make recommendations for changes in your medications to discuss with your physician so that you can avoid medications that might make your existing hearing problems worse.

How will the audiologist test my hearing?

After your audiologist talks to you in depth about your medical history, the next step is usually a hearing test. As we discussed in Chapter 2, the hearing test should be done in some type of sound-treated area. This can be a sound-treated room or just a small enclosure in the room, but it

should be a quiet area. Sound-treated rooms are generally metal walled rooms, looking something like a walk in cooler you would see in a grocery store. Background noise should not bleed through to distract you or cover up the sounds that the audiologist is trying to test.

Once you are in the sound-treated area, you'll be asked to put on headphones. When we're doing a hearing test, we really are looking to test your hearing at its best, not at its worst. Which is why it is so important to test in a very quiet environment free from background noise. We want to find out the very softest sounds that you can hear.

The audiologist will usually start by having you listen to detect very soft beeps. Beeps are very reliable and repeatable sounds. If you take a hearing test in my office, I know that the beep portion of the test will match up with the same part of the test if it's done in an office down the street and with tests I might give you later on. This consistency is important for correct diagnosis.

You'll listen to beeps of different pitches. (Think of a pitch as being a note on the piano.) The audiologist will ask you to either raise your hand or press a button every time you hear a beep. The beeps will get softer and softer until the point where you can just barely detect that beep (pitch) 50 percent of the time, which is called your *auditory threshold*. Your auditory threshold is the very softest sound that you can hear. It shows the lowest limits of your hearing when the environment is perfectly quiet.

Once we've determined the softest sounds that you can hear for a wide range of pitches (your auditory threshold), the next step in the hearing test is to find out how well you can understand

speech and words. After all, the reason you're looking into hearing aids is most likely that you're not happy with how you hear speech.

We test to determine the softest level of speech that you can hear. During this part of your hearing test, you will be given two-syllable words and be asked to repeat them. Every time you get a word right, we'll turn the volume down and give you another word. If you get that one right, we'll turn the volume down again until we find the softest point where you can understand those two-syllable words 50 percent of the time. That's called your *speech reception threshold*, or SRT. In short, your SRT tells you how far away from people you can be and still hear them.

The next type of speech testing that we do in a complete hearing test tells us your *word recognition score*, sometimes called your *speech discrimination score*. We typically have you listen to speech at your most comfortable volume. We'll find out what level sounds loud enough for you—for example, where you would set the volume on the TV if you were listening to speech. At that most comfortable volume, in that perfectly quiet room, we'll give you a list of words to see how many of them you can understand. This will tell us what percentage of speech you understand when everything is perfect—when you're hearing speech at your best with no competing noise.

Another type of test that you may have is a tinnitus evaluation. Tinnitus is any sound that you perceive that is not in the world around you. The most common type of tinnitus is a high-pitched squeal. All of us have left a noisy event and noticed that our ears were ringing. That is temporary tinnitus, but almost 20 percent of the population has that type of sound much of the

time. They might have ringing or other noises in their ears all the time, or it may recur intermittently.

Tinnitus can sound like any number of different things: ringing, buzzing, roaring, hissing, thumping, crickets, dinging, or even music. If you have tinnitus that bothers you significantly, your audiologist may also want to do a tinnitus evaluation. Unfortunately, there is no way to put a microphone in your ear and determine how loud your tinnitus is, because it's not a real sound in the vast majority of cases. No one else can hear it. Tinnitus is typically caused by neurologic or biologic activity deep in your ear or in your brain.

If you report tinnitus, you may be asked to complete a questionnaire about the effect the tinnitus is having on your life. You will often be asked to judge how loud the tinnitus is and what pitch it is. This information can often give the audiologist some information about whether a hearing aid or sound therapy device will reduce the impact of tinnitus on your daily life.

When they get a hearing aid, approximately 60 percent of people report that they have some relief from their tinnitus, and some patients even report total relief from their tinnitus while wearing hearing aids. Other devices also are available to give relief from tinnitus and are discussed in Chapter 6. The tinnitus evaluation should give you some idea of whether you are one of those people who get some relief.

Your audiologist may also do a speech-in-noise test. Almost everyone with hearing problems reports that their biggest problem is understanding speech in a noisy environment, like in a restaurant. This is not surprising, since a noisy place is also the most difficult listening

environment for people with normal hearing. Sometimes, though, we want to figure out if you are having a little difficulty in hearing in noise or if you having truly debilitating difficulty. In this case, it can be useful to see what amount of speech you can hear in the presence of noise.

The most common test puts multi-talker babble in the background—similar to a restaurant or cocktail party. If the test reveals that you have a significant amount of difficulty understanding speech in noise, it indicates that you'll need the most aggressive noise reduction strategy possible in the hearing device. In extreme cases, supplemental listening devices may be helpful in conjunction with your hearing aids to help overcome the problem of background noise.

Usually, the last part of your hearing evaluation will be bone conduction testing. This should be part of every diagnostic hearing test. We put a small vibrator box typically behind your ear, where it rests on the bony portion of your skull. Bone conduction is another test that typically uses beeps or tones. We vibrate your skull, which shakes the fluids in your inner ear. You hear it as sound, though, rather than feeling it as a vibration.

When we do bone conduction testing, we are able to get around the structures of your ear canal, your ear drum, and the bones and muscles of your middle ear, to actually test what your inner ear is hearing directly. Bone conduction testing helps us find out what part of your ear is experiencing difficulty.

A big difference between the results of your bone conduction testing and your earphone testing is usually a good sign. It usually means that an ENT physician may be able to improve your hearing loss with either medication or surgery. When there's *abnormal bone conduction*—when the results of the bone conduction and earphone tests are very different—the most common disorders that show up are earwax plugging up the ear and ear infections filling the middle ear with fluid. Both of these conditions are easily correctable medically. You may not even need hearing aids to fix the problem.

As a side note, your physician may do a modified bone conduction test using a tuning fork holding a tuning fork to your ear and then putting it behind your ear against the bone. If you're able to hear it when it's resting on the bone but not when it's free floating in air, it shows that you have abnormal bone conduction. This is a very basic test, but it can give your doctor enough information to know whether you have a hearing loss, what kind of loss you may have, and whether it is significant enough to warrant having a discussion of possible hearing devices. The tuning fork test is a good first step, but it is really designed to get you in to see your audiologist for a full diagnostic hearing evaluation.

After you have taken the tests, your audiologist should take the time to sit down with you, explain the results, and make sure that you understand what it all means. Most of your test results will be in a graphic form called an *audiogram*. Your audiogram is so important in your hearing aid success that I've devoted the next chapter, Chapter 4, to a discussion of how to read it.

But why do I need a hearing test before getting hearing aids?

Decades ago, hearing aids were simple amplifiers. They'd just make everything louder. They'd make loud sounds louder and soft sounds louder. They'd make low-pitched sounds louder and high-pitched sounds louder. The problem was that those hearing aids were a miserable failure.

They didn't take into account an individual's unique hearing loss. If you have normal hearing in the low pitches and terrible hearing in the high pitches, you need a hearing aid that will address those two sounds differently. You want a hearing aid that won't amplify low-pitched sounds at all but will amplify high-pitched sounds very aggressively. As audiologists, we don't know what you need to ask of your hearing aids, though, unless we do a hearing test.

Sometimes patients come into my office and say, "I've resisted having hearing aids for a while because I tried on my husband's hearing aids and they sounded terrible." The patients were well meaning; they thought that they would get an idea of what hearing aids could do for them by trying on any hearing aids. These devices, though, are very similar to eyeglasses. They are fit by prescription and are individualized to your individual hearing loss and your own neurology. You would never try on your friend's eyeglasses, realize that they made you nauseous, and then decide that you must not need eyeglasses.

Similarly, you'd never try someone else's blood pressure medication to see if you needed medication for high blood pressure. You need health care that is individualized to you. You need to try a hearing device that is made for you, not for someone else. You have to try a device that is actually fit to your individual needs to see if it works for you.

In addition, most states require by law that you have a hearing test prior to getting hearing aids. The hearing test must be up to date as well. In my home state, you're required to have a hearing test within six months of ordering hearing aids. One of the main reasons an audiologist is required to do a hearing test is for your safety. We are not just looking at whether hearing aids are appropriate for you. We're also finding out whether any medical treatment might be necessary. If an audiologist's testing raises any medical red flags whatsoever, we require you to see an ear, nose, and throat physician prior to considering hearing aids. If there's a way to fix your hearing problem through medication or surgery, it is almost always preferable to getting hearing devices. So we want to rule that option out for you, and the only way to do that is with a hearing test.

Please take the time to read Chapter 4 before your hearing test. You'll find your visit to the audiologist much more interesting and helpful if you know a little about the tests you'll have ahead of time.

What will happen after my hearing test?

Once you know the results of your hearing test, what you really want to know, of course, is what can be done about any problems that showed up. You waited a long time before making the appointment to go in and talk to the audiologist; you already know you have difficulty understanding speech. What you want to know is how to proceed.

At this point, your audiologist should be able to give you a good idea about whether your hearing loss has a good chance of being medically corrected. There are a number of medical red flags that always require a referral from an audiologist to an ear, nose, and throat doctor. Your audiologist will be able to determine whether an ear, nose, and throat doctor has a reasonable chance of treating your hearing problem.

If it looks like your hearing loss is permanent, though, the audiologist will almost certainly start discussing hearing devices as a treatment option. You should expect to talk with the audiologist

about different styles of hearing aids, different costs of hearing aids, and how the hearing aids will perform in different listening environments. Your audiologist should let you know what performance and level of success you will get if you decide to spend more or less money.

Your audiologist should explain the pros and cons of each device and help you figure out what device is right for you. You may even have the chance to listen to a device in the office. Ask the audiologist to program up a device to give you some idea about what it would sound like. Not all offices are equipped to do this, but if the office is able to demonstrate a hearing device, it gives you a massive amount of information before you make the decision on whether to buy a hearing device or not. You will know immediately whether things sound more clear, whether they sound more comfortable, and whether speech is easier to understand.

Once you've had your hearing test, talked about your device options, and settled on a budget, your audiologist can order a device for you for a trial period. Depending on what device you have chosen and how severe your hearing loss is, you may or may not need to have an impression taken of the shape of your ear.

An ear impression is a simple and painless process. When we need to custom-make the shape of a hearing aid to fit the ear tightly, which is done more frequently for severe hearing losses, it takes about five minutes. We place a small sponge into the ear canal and then fill the ear with a putty. The putty hardens up in about five minutes, and when we remove it we have a casting of the exact shape of the inside of the ear. We then use this to create a hearing aid that fits comfortably with little risk of falling out, even during physical activity.

What will I do at my hearing aid fitting visit?

On your next visit, the audiologist will program your hearing device for you and begin to teach you how to use it. This visit is called a *hearing aid fitting*.

Be patient with yourself at this visit. You're going to be asked to learn a lot of new skills, the most challenging of which is putting something into your ear. It doesn't sound that difficult, but ever since you were a child you've been told not to stick anything smaller than your elbow in your ear. Now somebody is going to tell you to do just that. Realize that things will feel a little clumsy for the first few days but that it all gets better with practice.

How can I tell if my hearing aids are working well?

Once you've learned the basics of how to operate your hearing aids, you'll have the opportunity to go out and try them in the real world. Your audiologist will suggest that you go out into the real world with your new hearing devices for one to two weeks and then check back in to discuss how things went.

It's very easy to try out a hearing aid in a medical office and think that things sound wonderful. However, I don't think I've ever had a patient come in and tell me, "I want a hearing aid because I want to hear better in your office, Eric." Most people want to hear better in their home, in restaurants, and in the workplace. It's absolutely essential that you take this device out and put it to the test in your real world.

During these first few days of hearing aid use, be sure to keep a list of things that you have really liked and things that you'd like to improve. Keep notes to remind yourself what to tell the audiologist. If there's something specific that's giving you problems, be as specific as possible. For example, you might tell the audiologist, "I generally do pretty well, but I can't hear my wife when she's more than 10 feet away." That type of information is absolutely essential to help the audiologist make adjustments so that you can be a successful hearing aid user.

I love when a patient comes in and says, "I'm doing great, but when I rustle the newspaper, it sounds like firecrackers." When somebody is that specific, we can make massive improvements in the sound quality right away. Don't ever feel like you are complaining or being a burden to your audiologist. Audiologists are here for you; you're not here for them. As an audiologist, the worst thing I ever hear from a patient is that they've been leaving their hearing aid in a drawer. If there's something that's bugging you, go in and talk about it. It's your audiologist's job to fix it. You are in charge.

Chapter 4

Do I Even Need Hearing Aids? Understanding Your Hearing Test

Early in my career, a woman, Nancy, visited my office for a hearing test. Her children had become alarmed about her hearing, to the point where they didn't want her driving anymore because they didn't feel it was safe. She couldn't hear emergency sirens coming up behind her; she couldn't hear traffic around her. Her family was concerned that she wasn't safe on the road.

The bigger problem, though, was that Nancy didn't believe that she had a hearing loss. She didn't understand that it was time to seek help. When I saw her for her hearing test she had a severe hearing loss. She could hardly hear any speech at a conversational level. Everyone around her had to speak up or even shout for her to hear them.

In my office, she went into the testing room, and we completed the tests. When she came out of the soundproof area, she informed me that she had done perfectly on the hearing test—that she had heard every beep that I presented to her and had pressed the button every time she heard the beep. The problem was that she didn't know what she didn't hear. She was correct that every time she heard the beep she pressed the button, but she didn't hear approximately 70 percent of the sounds at all. She had no idea they were even occurring.

I then gave Nancy a trial set of hearing aids and presented her with a list of words at a normal conversational level. With the hearing aids on, she understood 80 percent of the words. With the hearing aids off, she understood 0 percent. But she still insisted that she did not need hearing aids. She literally could not register that she was doing better with the hearing aids on.

The fact is that we do not know what we do not hear. We assume that our hearing is perfect, because we do hear a portion of speech. The only way to truly know whether you have a hearing loss that affects your speech understanding is to go in and get your hearing tested. Hearing loss isn't only about hearing. It is also about understanding.

In this chapter we talk a lot about a test result called an *audiogram*. It is my hope that as you learn what this test result means, you'll be better able to have a conversation with your audiologist about whether it is the right time for you to consider hearing aids.

How does my audiologist use my audiogram to find out about my hearing?

As we discussed in Chapter 3, your audiologist will test your hearing while you are in a soundproof booth. The results will be plotted on a graph, called an *audiogram*, which the audiologist uses to get a good deal of information about your particular hearing loss.

Your hearing test reveals several things that can be incredibly helpful in understanding what sort of hearing loss you have and how much benefit you'll get from hearing aids. You'll remember from Chapter 3 that your audiologist will determine your auditory threshold, speech reception threshold, and word recognition score. We'll go through each of these here and learn how to interpret your audiogram to see how you fared in each area, what that tells about the kind of hearing loss you have, and the kind of hearing aid that will therefore work best for you. We'll also learn what your bone conduction test reveals.

What does my auditory threshold mean to me?

When we talk about hearing loss, certain categories describe how severe the loss is. We say a person's hearing is normal or that the hearing loss is mild, moderate, severe, or profound, in order of worsening hearing. Your audiologist will use an audiogram to determine the severity of your hearing loss for various pitches by determining your auditory threshold.

The audiologist will first ask you to listen to beeps of different pitches. (Again, think of a pitch as a particular note on a piano.) As described in Chapter 3, the beeps will be made softer and softer until the point where you can just barely detect a particular beep 50 percent of the time. This is your auditory threshold for that pitch. (Remember, your auditory threshold is the very softest sounds that you can hear.) It shows the limits of your hearing in a perfectly quiet environment.

The audiologist will plot the loudness for each beep (or tone) on an audiogram. You will be tested for very deep, bass tones and for very high, shrill, squeaky tones to find the softest sounds that you can hear for a wide range of pitches. We select those particular pitches because they are the most important pitches for understanding speech. In general, the low-pitched, bass sounds that we test correspond very well to vowels. They also correspond very well to nasal consonants like m, n, and ng. The high-pitched sounds, in contrast, tend to correspond very well to breathy consonants, particularly those produced right in the front of the mouth at the lips and teeth. Sounds like s, f, t, and th tend to correspond with the highest-pitched sounds we test.

By measuring the softest sounds you can hear at all the different pitches, we can reliably predict what speech sounds you can and cannot hear. That tells us what kind of an effect the hearing loss is likely to have on your daily life.

For example, if you can't hear any of the speech sounds, it's easy to predict how that will affect your daily life: you are going to feel like you're just not hearing anything. Most people with hearing loss, though, have relatively good hearing in the low pitches but significant hearing loss in the high pitches. If that's your case, the hearing test will tell us that you typically will hear people but not understand what they're saying. This is a very different problem to solve than when a patient just can't hear anything.

Take a look at the audiogram provided here. The audiologist has plotted the patient's threshold for each pitch (the softest level of sound that allows the patient to just barely detect the pitch) on the graph. The "o" symbols represent thresholds for the right ear, and the "x" symbols represent thresholds for the left ear. If you were able to hear very soft sounds, your thresholds would be located at the top of the graph. The lower your marks fall on the graph, the louder that the sounds have to be for you to hear them. The numbers on the left side of the audiogram are a measure of loudness or intensity. The numbers refer to decibels (abbreviated dB). Larger numbers indicate louder sounds. A 100 dB sound is much, much louder than a 20 dB sound. You are able to hear only that part of the speech area that falls below your thresholds on the audiogram.

If you look at the audiogram, you will see where on the graph each kind of hearing loss falls. In general, if your auditory threshold is between 0 and 20 dB, your hearing is categorized as

normal. If your hearing falls between 21 and 40 dB, it's categorized as mild hearing loss. Moderate hearing loss generally falls between 41 and 70 dB. Severe hearing loss falls between 71 and 90 dB, and a hearing loss greater than 90 dB is categorized as profound. These different categories can be incredibly important for understanding how your understanding of speech will be affected by hearing aids.

Jet engines	140
Shotgun	130
Rock Concert	110-140
Power saw	110
Garbage truck	100
Subway/motorcycle	88
Lawnmower	85-90
City traffic	80
Washing Machine	75
Vacuum cleaner	70
Normal conversation	50-65
Refrigerator hum	40
Whisper	30
Rustling leaves	20
Normal Breathing	10

A mild hearing loss causes problems understanding speech when the listening environment is not perfect. People with mild hearing loss can usually understand conversation in a quiet room if they are facing the person who is talking. Many people with mild hearing loss find that hearing aids give them that little bit of added help they need to communicate effectively.

Moderate hearing loss causes difficulty understanding speech in many situations. People with moderate hearing loss may always hear that people are talking, but they will often have trouble understanding speech without wearing hearing aids.

Severe hearing loss and profound hearing loss are devastating to speech understanding unless treated with hearing aids. People with this level of hearing loss frequently find hearing aids extremely helpful, but they will always communicate with difficulty, even when wearing the best hearing aids. If the damage to the ears is so profound that hearing aid use is not enough to allow speech understanding, surgically implanted devices may be discussed as an option.

What is the "most important speech area" that is indicated on my audiogram?

When you're looking at your audiogram, it's important to realize that not every speech sound is the same. The sounds fall on different areas of an audiogram: some speech sounds are low pitched (of low frequency) and some are high pitched (of high frequency). Vowels, nasal consonants, and r and l sounds tend to fall on the left-hand side of the audiogram in the most important speech area. Breathy consonants such as s, f, t, and th tend to fall on the right-hand side. If you look at the audiogram in the figure, you will see that some of these speech sounds fall above the plotted line—above the x's and o's—and other speech sounds fall below the plotted line.

The audiogram pictured here shows the most common type of hearing loss. This person has relatively good hearing for deep, bass, low-pitched sounds, and relatively impaired hearing for shrill, squeaky, high-pitched sounds.

On the left-hand side of the audiogram, the plotted thresholds (the x's and o's), fall further up toward the top of the page. On the right-hand side of the audiogram, the plotted thresholds fall lower on the page. Any of the speech sounds that fall below that plotted line will be audible. Any of the speech sounds that fall above that plotted line will not be heard by this patient.

In this case, the patient may think that their hearing is fine. Because they have normal or near normal hearing in the low pitches, they hear vowels just fine. The problem is that they do not understand speech well; note that most of the consonants fall above the plotted line, meaning that the patient cannot hear them. So they will hear one portion of speech but not another portion of speech. When somebody says a sentence such as "I went to the zoo yesterday," the patient may only hear "I went o e oo ye-er-ay". When all of the high-pitched consonants that they're not hearing are filtered out, that's what is left.

This is a very typical kind of hearing loss, and this is a very typical experience that people with normal hearing at some pitches and poor hearing at other pitches will have. They will hear, but they will not understand. Still, they will think that their hearing is fine. They hear "I went o e oo ye-er-ay," so they think that's what the person said. People with this type of a hearing loss often say that it sounds like people are mumbling or not talking clearly. The fact is, though, that they are trying to piece together a jigsaw puzzle with 30 percent of the pieces missing. It's a very difficult thing to do, and that's where hearing devices can help. Hearing aids can actually

amplify those high-pitched consonants to a point where you can hear them again, dramatically increasing clarity.

How does my speech reception threshold affect my hearing and how my hearing can be helped?

As we discussed in Chapter 3, once we've determined the softest sounds that you can hear for a wide range of pitches (your auditory threshold), the next step in the hearing test is to find out how well you can understand speech and words. We test to determine the softest level of speech that you can hear—your speech reception threshold, or SRT.

We determine your SRT for a couple of reasons. First, we do it to ensure that we're getting an accurate test result. If we look back at your beep testing and average your auditory thresholds at 500 Hz, 1000 Hz, and 2000 Hz, the average of those pitches should be approximately equal to your speech reception threshold. If we see a big difference between your speech reception threshold and the average of your auditory thresholds, we have to suspect that we aren't getting accurate test results. We need to go back and find out why. Perhaps the instructions weren't clear, or there may be a financial reason to exaggerate hearing loss, such as when a lawsuit or disability claim is being made.

The other reason your speech reception threshold is useful is that it tells us how loud speech has to be in order for you to understand it. In short, your speech reception threshold gives us an idea of what your listening distance is—how far away from people you can be and still hear them. Most people are able to understand speech at 20 dB or less. Generally, 0 to 20 dB is considered a normal speech reception threshold. If you have a speech reception threshold between 0 and 20, you should be able to understand speech at a normal conversational level, even all the way down to whispered speech, without much difficulty.

We often use the speech reception threshold to figure out whether it's time for you to get a hearing aid. If we see that your speech reception threshold is normal or near normal, it may be okay for you to wait a little bit before taking the plunge into hearing aids. But if we find that your speech reception threshold is much higher than normal, it's certainly time to get hearing aids. You've waited too long.

A speech reception threshold between 21 and 40 dB is considered a mild hearing loss; this is the point where most people start to notice an effect in their daily lives. At this point, you'll tend to find that your hearing is pretty good if you're face-to-face with the speaker and in a quiet environment. And even if somebody's back is turned or if they're more than 6 or 8 feet away, as long as the room is quiet you will probably still understand most of what's being said. But at this point you'll start to notice that your primary complaints are that listening to speech in background noise is becoming significantly harder. Restaurants, group meetings, and family gatherings may become more frustrating. And it may be that it's not so much that you don't understand what's being said, but that you misunderstand what's being said. This is the point where people start to talk about being afraid that they will respond inappropriately or answer a question that wasn't even asked. When people are at this point, they talk about being afraid of being laughed at in a social setting because they thought one thing was said when actually it was something completely different.

When your speech reception threshold is between 41 to 70 dB, you're categorized as having a moderate hearing loss. When your hearing loss has progressed to this point, you are heavily reliant on optimizing your listening environment. At this point, you really will only be an effective communicator if you're face-to-face and within 6 to 8 feet of the speaker. People with a moderate hearing loss will say that they cannot hear someone from the next room or cannot understand people whose backs are turned. Background noise makes hearing completely impossible, but hearing is fine as long as it's a face-to-face conversation.

That's the insidious thing about hearing loss. In some listening environments you may do well, but you won't in many others. By the time you have a moderate hearing loss, you are so dependent on trying to find the optimum listening environment that communication during most normal activities is unsuccessful. By the time you have a moderate hearing loss, you absolutely should be investigating the option of hearing devices to improve your situation.

If your speech reception threshold has risen to the point where it's between 71 and 90 dB, you're categorized as having a severe hearing loss. At this point, the hearing loss is impossible to ignore. Without the use of hearing devices, you will be a nonfunctional communicator. People will have to get right up to your ear and raise their voices; they may have to shout. You may be increasingly reliant on lip reading at this point unless you opt to get assistance in the form of hearing devices. Very few people will tolerate a severe hearing loss; most request immediate help.

Once your hearing loss has progressed to the severe level, there are increasing limitations on what the hearing aid can do for you. With severe hearing loss, we tend to see reduced word

recognition scores, which reflect how clear the speech that you hear is. Hearing devices will absolutely improve your ability to carry on a conversation, but as we dip down into the severe range of hearing loss, we will find more and more limitations on the ceiling of your performance, such as how effective you can be in challenging environments such as restaurants.

Once your speech reception threshold is greater than 90 dB, you're categorized as having a profound hearing loss. At this point, hearing aids can still provide some benefit, but the clarity of speech will almost always be significantly compromised. Even face-to-face conversation will require the use of visual cues such as facial expressions and gestures. People with profound hearing loss often consider using sign language as a possible supplement to what they're hearing. If your hearing loss has progressed to a profound state, it may be time to go in and talk about surgically implanted devices such as a cochlear implant if you want to maximize your ability to hear and understand speech. We discuss these devices in greater depth in Chapter 6.

Why is my word recognition score so important?

Your audiologist will also give you a word recognition test or speech discrimination test. To determine your score on this test, you'll be asked to listen to speech at your most comfortable volume. We'll find out what sounds loud enough for you—for example, where you would set the volume on the TV if you were listening to speech. At that most comfortable volume for you, in that perfectly quiet room, we'll give you a list of words to see how many of them you can understand. This tells us the percentage of speech you understand when everything is perfect—when you're hearing speech with no competing noise.

To learn why your word recognition score is so important, let's summarize what we've learned so far. When you have a hearing loss, you typically lose volume and lose clarity. Your speech reception threshold tells you how much *volume* you have lost. Your word recognition score tells you how much *clarity* you have lost. If you have a hearing loss, you probably do not understand 100 percent of speech even when it's at your most comfortable volume. This happens because you probably have little tiny "blind spots," or distortions, in your hearing. This is important to understand when you get hearing aids. Hearing aids will not just make things louder, and then everything will be perfect. You'll need to have realistic expectations about what hearing aids will really be able to do for you in the real world.

The word recognition score is the most important predictor for how you'll do with hearing aids in the real world. If I know that in a perfect listening environment, your absolute best word recognition is 60 percent correct, that tells me that no matter what hearing device I give you, you're going to have to use other strategies to understand speech 100 percent of the time. You may have to use your knowledge of grammar and context to understand language; you may have to use facial expressions or watch gestures to help fill in the gaps.

The word recognition score is incredibly important because if I can tell you in advance what the limitations of your hearing aids will be, you will be much, much happier with them. For example, if you know that your maximum word recognition is 60 percent, you won't have unrealistic expectations that a hearing device is going to allow you to understand 100 percent of speech. A hearing device should make things much, much better, but knowing the limitations is very empowering.

In short, the speech reception threshold tells us how loud speech has to be in order for you to hear it. Word recognition tells us how clear that speech will sound to you once it is loud enough for you to hear it.

Why do you do a bone conduction test?

Hearing loss can be conductive hearing loss, sensorineural hearing loss, or mixed hearing loss (a combination of the first two). Chapter 6 describes these kinds of hearing loss in detail.

Essentially, a sensorineural hearing loss is an electrical problem where the vibration of sound is not converted to an electrical signal that your auditory nerve can send to the brain. Sensorineural hearing loss is located deep inside your inner ear or in your auditory nerve or brain.

In contrast, a conductive hearing loss is a mechanical problem where the vibration of sound is being reduced. Conductive hearing losses are normally caused by problems with the ear canal, eardrum, or bones in the middle ear. These structures are much closer to the surface, and surgery or medication is often an option with conductive losses.

A bone conduction test gives your audiologist a reasonable idea of what type of hearing loss you have. If it turns out that you have conductive hearing loss, your audiologist can refer you for medical treatment and very possibly a cure.

For your bone conduction test, what you need to know is that conductive hearing loss is a mechanical problem, and it is generally easy to treat medically. In this kind of hearing loss,

sound is being damped down. The sound vibrations do not have a clear path through the ear canal or are not getting efficiently passed through the bones of the middle ear. The most common cause is an earwax blockage in the ear canal. If you have a plug in your ear, the vibrations of sound can't get to the eardrum to move the eardrum and the bones of the middle ear. Another common cause of conductive hearing loss is ear infection. In this case, there will be fluid behind the eardrum, and the pressure of that fluid will keep the eardrum from moving normally.

Conductive hearing losses are generally easy to identify and easy to treat medically. If there's a problem that's keeping the structures of the ear from moving normally, a physician can correct that, either through medication or surgery. If the problem is a plug of wax in the ear, a good physician can remove it quickly and easily. If there's fluid behind the eardrum that isn't allowing the eardrum to move normally, that fluid can be drained off either by curing the infection that's causing it through antibiotics or by putting a small hole in the eardrum to allow the fluid to drain out. Once the fluid is gone, the eardrum will move normally again.

In children, the middle ear is sometimes ventilated to keep fluid from building up. A physician will simply make a small hole in the eardrum and place a tube through it, allowing fluid to drain out as it builds up. This keeps the middle ear dry and, it is hoped, avoids ear infections in the first place.

When we look at the audiogram, we can determine whether there's a conductive hearing loss by examining whether there's a difference between air conduction and bone conduction. This can be seen on the audiogram by comparing the x's and o's with the small brackets. The small

brackets indicate bone conduction. Bone conduction is determined by putting a small vibrator box back behind the ear and shaking or vibrating the skull. This bypasses the mechanical structures of the ear and stimulates the auditory nerve directly.

If your bone conduction scores (the brackets) are almost identical to your air conduction scores, (the x's and o's), then you do not have a conductive hearing loss. You have sensorineural hearing loss. In this case, hearing devices are the preferred treatment (see Chapter 6). But if there's a big difference between the x's and o's on the graph and the brackets on the graph, then we know that you have a conductive hearing loss. In that case, your audiologist will refer you to a medical doctor as part of your treatment. You may still need hearing aids, but medical treatment is the first course of action

What do blind spots on my audiogram mean for my hearing aid success?

If your hearing loss has progressed to a certain degree, usually into the profound range, it's very possible to have blind spots, often called *cochlear dead regions*, in your hearing. This means that what we measure on the audiogram is not real hearing. What we actually may be measuring is distortion caused at a different pitch than we think we're measuring.

If we see high-pitched hearing loss on the right-hand side of the graph that is worse than 80 to 90 dB, we start to suspect that there may be a blind spot in the hearing. What may be happening is that as we're putting enough sound behind a high pitch to stimulate that 80 to 90 dB response on the audiogram, we may actually be creating a lower-pitched distortion that you detect but not as a clear, high-pitched sound.

If you're in a soundproof room taking a hearing test and you have unusual perceptions of a sound—maybe we give you a beep but it sounds like crickets, static, or chirping, but definitely does not sound like a beep—that is a major red flag that you may actually have a blind spot in your hearing, a cochlear dead region. Knowing this is very important in selecting a proper hearing device for you. No matter how loud we make a speech sound, it may not be loud enough for you to hear it at all if you have a blind spot or cochlear dead region. This can have big implications for what type of device we want to proceed with.

If your hearing loss is in the severe or profound range, be sure to ask your audiologist what realistic outcomes you can expect from a hearing device. Also ask if there are any specialty devices that you should consider to maximize your speech understanding. In Chapter 6 we discuss hearing devices that can specifically address blind spots in hearing.

Chapter 5

Why Is Hearing Loss Harder to Correct Than Vision Problems?

When patients come to me asking for help with their hearing, most people think of this process as being very similar to getting eyeglasses. It is not.

For example, one patient, Jake, came to me with a shoebox full of hearing aids. He had tried five or six sets of hearing aids in recent years, and he had been disappointed by every set. The reason is that fitting hearing aids is very different than fitting eye glasses. When you're fitting eye glasses on someone with nearsightedness, all you have to do is focus the image on the eye, and everything is great. That is because all of the cells that send the visual signal to the brain are still alive and working once light hit them.

Jake actually had a true blind spot in his hearing: there were certain high-pitched sounds that he would never be able to hear no matter how loud we made those sounds. And every attempt at hearing aids in the past had been with very traditional hearing aids, which just tried to make the high-pitched consonants loud enough for him to hear. This was a miserable failure, because it was not possible to make him hear these sounds.

What I had to do with Jake was to give up the idea that all I had to do was amplify or focus the sound for him. I had to address the fact that hearing is different than vision— that he in essence had a blind spot in his hearing that we had to work with. We were able to find devices that could actually shift, or warp, sound so that they brought the high-pitched sounds into a frequency range that he could hear. This allowed him to hear sounds like *s*, *f*, and *th*, although they would sound different than he remembered before he had a hearing loss.

In addition, we talked about compensatory strategies he could use to allow him to supplement what he wasn't hearing with his vision, with his knowledge of English grammar, and with context. By treating him as a whole patient, rather than just a set of ears, I was able to get Jake to be highly successful with his new hearing aids.

Why do my glasses make things perfectly clear, but I always hear from my friends that hearing aids don't do that?

Hearing aids are not like glasses, because hearing problems are different than the vision problems most people experience. When most people think of vision problems, they think of being either nearsighted or farsighted. Things are just a little out of focus, but the eyes are generally in good health. They don't have blind spots along with their nearsightedness. There's no distortion. There are no holes in the retina. Most people have healthy eyes that can receive light and send that signal up to the brain. All they need to do is get glasses that can focus the light properly.

Unfortunately, that's not the case with hearing loss. With hearing loss, we typically see a loss of the cells in the inner ear that are responsible for sending a signal up to the brain, called hair cells. Essentially, this means that most people with hearing loss have hundreds of tiny blind spots in their hearing. Even if we make things loud enough or amplified enough, there just is no cell to convert that sound to a language that the brain can understand.

Hearing loss, then, is much more like nearsightedness plus macular degeneration, a condition that causes blind spots in your eyes. Macular degeneration is a vision problem that often affects

older adults in which some of the sensory cells in the eye are destroyed, causing blind spots. The patient with macular degeneration may have good peripheral vision on the sides but not be able to see anything straight ahead. Say a patient has nearsightedness and a blind spot in the middle of the vision. The eye doctor can correct the nearsightedness with glasses, but he cannot correct the blind spot. In this case, the glasses are more like hearing aids in what they can do: they can correct part of the vision problem but not the whole vision problem.

In hearing loss, you have a loss of outer hair cells, which means that you won't hear soft sounds very well. Sound is not loud enough. You'll need extra volume. In addition to that, you may have a loss of inner hair cells, which means there are tiny blind spots and distortions in your hearing. Hearing is not clear. Hearing aids are designed to correct the loss of outer hair cells by making the softer sounds of speech loud enough for you to hear. What hearing aids cannot correct is the loss of inner hair cells, which affect the clarity of your hearing.

When you lose your hearing, you tend to lose outer hair cells first. These are the cells that are responsible for helping you to hear soft sounds. When those go, you lose the ability to hear soft speech or the softer portions of speech, so you may not be able to hear a whisper at all. You will tend to still hear loud sounds just as well as you always did. So although you may not hear a whisper, a shout may still sound painfully loud.

This situation caused big problems with older technology hearing aids. The hearing aids that were designed in the 1970s and 1980s frequently operated as very simple amplifiers. They would make everything louder. They would make soft sounds louder, and they would make loud
sounds louder. That certainly could help sometimes, but the problem was that typically loud sounds sounded uncomfortable, but soft sounds still weren't loud enough to be audible.

Newer hearing devices can address this issue very effectively through the use of automatic volume control, which we discuss in greater depth in Chapter 7. For now, remember that modern hearing aids can deal with your hearing loss just for soft sounds, in ways we never could before.

Also, even if you have a substantial loss of inner hair cells, hearing aids should give you a significant benefit. You should never expect hearing aids to get you back to your 18-year-old ears—your pre-injury state—but you should expect them to be of significant benefit in most situations. If your inner hair cell loss becomes severe, though, you and your audiologist might need to talk about options in addition to hearing aids.

What happens if I don't treat my hearing loss?

As we discussed in Chapter 1, most people put off doing something about their hearing loss as long as possible, and unlike uncorrected vision problems, waiting until a hearing loss is impossible to ignore carries a significant risk.

If you get the correctly prescribed glasses at any time, you'll be able to see clearly right away. In contrast, if you've gone decades without doing anything about your hearing loss, it will probably take you some time to regain some neurologic and hearing function. When you have a long-term hearing loss, it takes a significant amount of stimulation to regain the brain's ability to understand speech.

If we don't actively use our brain's ability to hear and understand speech, we lose some of that ability over time. We hear with our brain. The ear is just a mechanism to change sound from a physical motion into an electrical signal that can be sent up to your brain. Once that electrical signal goes up the auditory nerve, through the base of your brain, and gets up to the higher language center of your brain, that's when you actually register that sound has been heard. In order for those neurons and cells to remain healthy, they need to have a steady diet of sound. If your auditory neurology does not get stimulation, it can atrophy. It can degenerate. It can become less able to understand the sounds that it receives.

If you have gone a year without treating your hearing loss, it will generally take a month before you get the full benefit of hearing aids. If it's been 24 years that you've been ignoring your hearing loss, strap yourself in for about two years during which your hearing will continue to optimize as you're wearing the hearing aids. **You should absolutely get benefit from your hearing aids from day one, but it may take a little bit of time to get to your final plateau of performance.**

And remember, there actually is a point of no return. When you have ignored your hearing loss long enough, you may never be able to get all of your hearing function back.

How does my hearing change with age?

Most people think that hearing loss is always age related. While it's true that hearing loss is much more likely to occur as you get older, that's primarily due to the fact that you are exposed to more things over time. You are exposed to more toxic substances, high noise events, and

medications; if you're 80 years old, you've had ten times the life experience—and damage to your hearing—as an 8-year-old, and that damage can add up over time.

When we go into tribal societies that are not in industrialized nations—say, a tribe in the the Sudan—and test the 75-year-olds, we find that their ears hear almost as well as the 25-year-olds in that tribe. They don't have the same degree of age-related hearing loss that we tend to have in our industrialized society.

Ears can remain very healthy late into life. However, there does seem to be an age-related neurologic decline in hearing. A number of studies have shown that between the ages of 70 and 75, our auditory brain stem, an area at the base of the brain that does some of the most basic neurologic processing of sound, loses a number of cells that are responsible for hearing rapid changes in pitch and loudness. Unfortunately, speech is entirely made up of rapid changes in pitch and loudness. This age-related change leads to subtle, hard-to-measure difficulty with hearing in background noise and difficulty with rapid speech, even in people with normal audiograms.

With this age-related neurologic decline in mind, it's important to understand that if you're in your 70s—even with the best hearing aids that can get you back to hearing the sounds as soft as you did when you were 20 years old—there are some limitations to what the hearing aid can do in background noise. They can certainly make it easier to hear in background noise, but I want to be very careful and caution you that it's important to have realistic expectations and to view your hearing aids as providing benefits, not perfection.

Are there special considerations regarding difficulty in correcting children with hearing loss?

Children with hearing loss are a special category. The critical period for learning speech and language is from birth to age two. Children who acquired their hearing loss before the age of two, then, will typically experience some delays in acquiring speech and language and will need support from educational professionals.

The most important consideration is whether a child will learn to speak or will learn language through sign language. This sensitive issue is very individual and is beyond the scope of this book. At this point, I would just like to say that if your child has a hearing loss prior to the age of two, please consult with deaf educators, speech and language pathologists, and your audiologist in person about what the options are to help maximize the child's ability to communicate either through speech and language or through sign language.

What can I expect from treatment for my hearing loss?

Very simply, what you should expect from treatment for your hearing loss is *benefit*. You should expect better speech understanding in almost all situations while wearing hearing devices. I'm going to be very careful here and stress that you should expect *benefit* or *improvement*. Hearing devices will not give you perfect hearing 100 percent of the time. You will still have situations where your hearing is not what you want it to be.

I generally tell people to think about it in the following terms. If you take all of your difficulties understanding speech together and cut them in half, that will be a reasonable expectation from

your hearing devices. If you take the number of times you ask for repetitions and reduce it by 50 percent, that is a perfectly realistic expectation of the benefit you'll get from your hearing aids.

Hearing aids will not make your hearing normal and will not completely eliminate your requests for repetition. The reason why is very simple. My hearing is better than normal. Through most of my life, my hearing has been so good it's unmeasurable. You can't even plot it on the audiogram. But I still ask people to repeat what they've said. With better than normal, off the chart, unmeasurably good hearing, I still have difficulty sometimes. Therefore, with a hearing loss and hearing aids, you will still have difficulty at times, too.

The most common situation in which people want improvement in hearing is amid background noise such as at a restaurant or party. An interesting study done a few years back asked a number of people, "How much speech do you think you understand when you're in a cocktail party setting?" People who did not use hearing aids said they understood between 15 and 20 percent of speech in that situation. People with one hearing aid said they understood about 35 percent of speech, and those with two hearing aids said they understood about 65 percent of speech. Those with normal hearing said they understood about 70 percent of speech.

So, when you choose to get hearing devices it's important to remember that they will not get you back to understanding 100 percent of the speech you hear, because no one hears perfectly. Even people next to you at a party with normal hearing feel like they are smiling and nodding and faking it 30 percent of the time. If you're eating out in a restaurant with your hearing devices and you feel like you're smiling and nodding and faking it only 30 percent of the time, that's a pretty good outcome. That means you're doing about as well as somebody with normal hearing.

What you should find is that when you wear your hearing devices, there's an *improvement*. I suggest that the first week you get your hearing devices, you go out to a restaurant and listen to half of the meal with the hearing aids in and the other half with the hearing aids out. Sit through the salad with the hearing devices on as you talk to your dining companion, then sit through the entrée with the hearing aids out, and you will immediately see whether there's a difference. When people do that, they tend to report things like, "Okay, when I was without the hearing aids, I understood 20 percent of the conversation. With them in, I understood 60 to 70 percent of the conversation." That is definitely a successful outcome.

The problem is that we get caught up in wanting 100 percent. We have actually forgotten that "normal" hearing is not perfect hearing. We actually want our hearing aids to get us better-thannormal hearing, which is not realistic.

So it is important to understand that hearing aids are designed to get you as close to normal as you can possibly get. Hearing well in very difficult listening environments, such as hearing somebody when you're in the back seat of the car and they're in the front seat, probably is not a reasonable expectation. Hearing your spouse talking from the next room while you're running water in the sink is probably not a reasonable expectation. However, it certainly is reasonable to expect improvement in almost every listening environment. If you understand 20 percent with the hearing aids out and you understand 70 or 80 percent with the hearing aids in, that's what we're going for.

I have the most trouble understanding my spouse. Why doesn't he just talk more clearly?

One of the most common requests my patients make for their hearing aids is to understand their spouse or partner better: "I can hear other people pretty well, but I can hardly hear my wife or husband at all." Spouse-to-spouse or partner-to-partner communication is different than any other kind of communication.

We have different conversational habits with our family members than we have with anyone else. We cut corners with our spouse that we would never cut with an acquaintance. We would never invite a buddy over to our house, walk out of the room, turn our backs on them, turn on the water, do dishes, and expect to carry on a conversation. But this is completely common between married couples. I'm not trying to place blame here, but it helps to be aware that this is something we do with the people we're around most of the time. We don't take the same care in talking to them that we take with our friends.

When you have normal hearing, you may or may not hear each other well. I guarantee, though, that if you have a hearing loss and you try to carry on a conversation from the next room with the water running, it's not going to work no matter which hearing aids you bought.

Having said all of that, even though we are sloppier in our communication habits with our spouse, difficulty understanding them can be the first indicator of a hearing loss. Because the communication environment and habits are more difficult, mild hearing losses will be much more obvious when you are communicating with your spouse in the home. If you're feeling like you understand everyone just fine, but not your spouse, that is a major red flag. It's time to go in and have your hearing tested.

Chapter 6

What Types of Help for My Hearing Loss Are Available?

There is a limit to what hearing aids can do. Hearing aids are amplifiers; they are not replacements for the ear.

Julie, a patient who came in to see me recently was in her fifties and worked in retail. She had a fairly severe hearing loss and already wore hearing aids. She came to me hoping to be able to hear better in those noisy stores, as well as on the phone. She hoped that the newer, high-tech hearing aids might allow her to communicate more effectively. Unfortunately, she had such a severe hearing loss and so much distortion in her ears that no hearing aid I put on her was going to allow her to hear clearly in those situations.

After evaluating her performance with the highest tech hearing aids on the market, we determined that she needed a different approach. We went through a long counseling session about a device called a cochlear implant, which is a surgical device that we talk about at length in this chapter as an alternative to hearing aids in cases of severe hearing loss. In this particular case, it made much more sense to refer Julie to a physician for a surgical consultation.

If you have a severe hearing loss, there are many types of treatments that we can talk about. Some of them are hearing aids, some of them are other devices, and some of them are simply education and strategies. You are an individual, and there is no one set way an audiologist can help you meet your goals. When you talk with your audiologist you can expect to have a discussion about what your particular goals are and what treatment options, out of all of the treatment options available, are best for you, and why.

As you read in Chapter 4, the type of help your audiologist can provide varies by the type of hearing problem you have.

I have conductive hearing loss. How can this type of hearing loss be treated?

Conductive hearing loss is a mechanical problem in your ear. To put it simply, parts of the ear are not moving. The air molecules in the ear canal may not be moving because of a plug of earwax, and thus they cannot cause the eardrum or bones of the middle ear to move so that you can hear. This type of hearing loss is the kind that is most treatable by medical therapies.

If you have a conductive hearing loss, your ear, nose, and throat doctor will be involved every step of the way.

The doctor will try to remove the impediment that is keeping the structures of the ear from vibrating. There may be a plug of earwax that can be removed or an infection that can be cleared up with antibiotics. In some cases, surgery may be used to either drain off fluid from the middle ear or to reconstruct bones that have been broken or degraded.

Conductive hearing loss can be medically corrected most of the time. However, in certain instances the conductive hearing loss will be permanent. In these cases, a couple kinds of hearing devices can help. The first type is a traditional hearing aid. Because the problem with conductive hearing loss is simply that vibrations are being damped down or softened, and a hearing aid makes vibrations louder, or bigger, hearing aids work phenomenally well with conductive hearing loss. Once we make sound loud enough, it typically will be very, very clear. People who have corrected a conductive hearing loss with hearing aids hear very similarly to someone with normal hearing.

There are times, though, when hearing aids do not work well with a conductive hearing loss. A chronic ear infection that causes fluid to drain out of the ear will cause a hearing aid to malfunction. People with

certain birth defects may not have a visible ear or ear canal, so sound cannot enter the skull. In these cases, a new type of device called a bone-anchored hearing aid is surgically implanted. It causes the skull to vibrate, which then vibrates the fluids of the middle ear. These devices can even help people born without an ear.

If I have a sensorineural hearing loss, how can it be treated?

Sensorineural hearing loss is caused by damage to the hair cells in the inner ear or damage to the auditory nerve. The inner ear, or cochlea, changes sound from a mechanical sound to an electrical signal; the auditory nerve carries that electrical signal up to the brain where hearing actually occurs. This damage commonly occurs due to exposure to loud noise, aging, and heredity.

Sensorineural hearing loss is treated with hearing devices, rehabilitation strategies, or both. Hearing devices include hearing aids (including assistive listening devices, personal sound amplification devices, and tinnitus instruments, all discussed in this chapter, as well as cochlear implants). Rehabilitation strategies might be lip reading, sign language, and active listening therapy.

Which treatment or combination of treatments that might work best for you will depend on your own particular hearing loss. Your audiologist will discuss the options with you and give you the information you'll need to decide what will help you the most. The rest of this chapter is devoted to these treatments for sensorineural hearing loss, since it is so common.

How do hearing aids help with a hearing loss?

For the vast majority of people with sensorineural hearing loss, hearing aids are the most effective kind of treatment available. The reason is that the first structures in the ear to be damaged are typically the outer hair cells. These are the cells that make sounds loud enough for the brain to interpret. Modern hearing devices selectively amplify soft sounds. They don't necessarily amplify loud sounds; they try to replace the function of those cells that are most commonly lost.

With the best possible fitting of hearing aids, you should feel not so much that you're getting a lot of volume or loudness, but that you're getting clarity. The hearing aids should selectively amplify whispered speech while letting shouted speech come through as if it's not amplified at all. The bottom line is that hearing aids should make soft speech audible, moderate speech comfortable, and loud speech tolerable. If we can do that, you're generally going to have good success with your hearing aids.

Are cochlear implants an option for me?

If your hearing loss is in the severe to profound range, there is a diminishing level of return with hearing aids. They may still be a significant help to you in understanding speech, but sometimes a surgically implanted device called a cochlear implant may be a better choice. If your word recognition score has dropped significantly—if at your most comfortable volume you can hear and understand only 30 to 40 percent of words—a cochlear implant may help you.

In a severe to profound hearing loss, decreased clarity occurs because you have lost inner hair cells in your ear. These are the cells that convert sounds to electrical impulses, which travel up to the brain. Once the inner hair cells are damaged or have died off, they don't grow back. This leaves you with tiny blind spots in your hearing. If you have enough deterioration of the inner hair cells, your clarity of speech drops dramatically because the sound is no longer being efficiently transduced into an electrical signal.

A cochlear implant seeks to get around this problem. This is a surgically implanted device. It consists of a wire with around 22 to 24 electrodes attached to it that is placed inside your inner ear. The wire is wrapped around the end of your auditory nerve, and the electrodes stimulate your auditory nerve electrically. This takes over the function of the inner hair cells. When the auditory nerve is stimulated electrically, that electrical impulse gets sent up to the brain, where it can be interpreted as sound or

speech. Even though the stimulation is electrical, your brain can interpret it as sound. Remember, converting sound to electricity is what a healthy inner ear does. The cochlear implant attempts to take over that process when the inner ear has been catastrophically damaged.

It's important to note that cochlear implants are not going to make you hear like you did with your 18year-old ears. After all, instead of using 15,000 cells in your ear, they use 24 electrodes to try to give you information about speech.

Patients often tell us that speech through a cochlear implant can sound cartoonish or robotic—almost like people are talking like Mickey Mouse or Donald Duck. Usually, people report that voices they're familiar with usually sound fairly natural, but new voices may not. Even though speech may not sound clear or natural with cochlear implants, though, it typically sounds clearer than using hearing aids with a severe to profound sensorineural hearing loss. And usually, speech understanding is improved with a cochlear implant versus hearing aids in people with a severe to profound hearing loss.

If you're thinking about getting cochlear implants, it is important to discuss the results with a cochlear implant team in your area. Cochlear implants involve major surgery. They can be an immense improvement, but it's beyond the scope of this book to take you through every aspect of cochlear implants. I encourage you to contact a local cochlear implant team in your area if you're concerned and your hearing loss is severe to profound. They may be a viable option that you haven't yet considered.

What if I have mixed hearing loss?

Mixed hearing loss is a combination of conductive and sensorineural hearing losses. Typically, the conductive portion of your hearing loss can be treated by a medical doctor. The sensorineural portion is treated by an audiologist.

My hearing aids are not enough when I'm watching television or listening to a lecture. Can anything help?

Not everyone gets as much benefit from hearing aids or cochlear implants as they would like. The most common complaints are "I have difficulty understanding in background noise," "I have difficulty understanding speech at a distance," and "I can't understand television very well."

Sometimes technology in addition to hearing aids can be of great help. *Assistive listening devices* connect your hearing aid directly to the sound source you want to listen to. For instance, many people come into my office and say, "I need better clarity for the television. I'm setting things up louder than my family would like."

In this case, we can put a wireless transmitter on the television to send the sound from the TV directly to your hearing aids. That allows you to turn your hearing aids into wireless headphones, essentially plugging your hearing aids directly into the TV. This process overcomes the loss of clarity that occurs as sound travels through the air from a distance—from the television to your hearing aids. It also overcomes any problems with background noise in the room. When you patch your hearing aid directly into the television, the only thing you're amplifying is the television. And that can make a massive difference in the clarity of the sound you hear.

We can also use this same idea in a lecture or classroom setting. If you're having difficulty hearing speech at a distance, you could have the lecturer wear a small wireless transmitter mic clipped on their collar to send the lecturer's voice to you wherever you are in the audience.

This process may sound a bit cumbersome, because you do have to ask the lecturer to wear a transmitter mic. However, it can be an amazing benefit for people who struggle with large lecture settings. The same approach can be used in colleges, high schools, and even churches. Most churches have some sort of

sound system already installed, and there may be a way to patch your hearing aids into that existing sound system. Your audiologist can tell you what to do.

Is there any way I can hear better on my cell phone?

Cell phones are frequently a source of complaints with hearing aids. Again, assistive listening devices can help.

Believe it or not, the best sounding telephones we've ever had in this country were probably designed in the 1950s. They were robust and heavy. The loudspeakers were driven by a high-quality magnet and powered by electricity. All of these things contribute to a better sound quality. Today, the emphasis is on making phones smaller and portable, and we lose some sound quality in the process. It's common for people to say, "I hear well on my landline phone, but I don't hear well on my cell phone."

A number of assistive listening devices can allow your cell phone to wirelessly transmit the phone call to your hearing aids with a much higher quality. These devices usually use a transmission medium called Bluetooth, which is built into most modern cell phones.

In addition, the phone call can be sent to both ears at once, if you're wearing two hearing aids. This allows you to use the best parts of hearing in both ears to interpret the phone call, which can be a massive improvement for many patients. Many people with hearing aids understand 40 to 50 percent of speech in each ear individually, but they understand 80 to 90 percent of speech when they use both ears together. Being able to transmit the phone call from your cell phone directly to both ears can frequently increase clarity dramatically. These specialized devices are typically designed for a specific brand of hearing aid, so each manufacturer has its own version. Your audiologist should be well versed in helping you pick the best cell phone assistive listening device for your needs.

I saw something called a personal sound amplification device at the drugstore. Is this something worth looking into?

At the pharmacy or an electronics store, you may have seen a display for *personal sound amplification devices*, or PSAs. Patients ask me about them fairly frequently. You need to know that these devices are not actually hearing aids; they are simple amplifiers that are sold over the counter, meaning without a doctor's prescription. If you read the package, you'll note a disclaimer saying that they're not designed for hearing loss. However, all of the large text on the package will certainly lead you to the conclusion that they can help with a hearing loss.

These simple amplifiers make soft sounds louder, and they make loud sounds louder. They are not acoustically tuned for your particular hearing loss. You will know a personal sound amplification device because it is typically priced anywhere from \$20 to \$200, whereas hearing aids typically are priced close to \$1000 or more. In addition, because they're not designed for hearing loss, PSAs are not subject to the same laws as hearing aids. And because they're not fit to prescription, they can be sold over the counter.

A good rule of thumb is that if you don't have to have a hearing test and you don't have to have a professional involved to look in your ear and tell you what your problem is, what you are buying is not an actual hearing aid. And it probably is not going to give you the results that you really need.

Is there something that can help me with my tinnitus?

If you have tinnitus, you know how bothersome it is. Tinnitus is any sound that you can hear that is not in the world around you. Tinnitus can sound like buzzing, roaring, ringing, hissing, or thumping.

If you don't need hearing aids, a tinnitus device may help. This device is designed for people with ringing or other noises in their ears. It won't usually get rid of the tinnitus, but it is designed to make the

ringing or other noises in your ear less annoying. They accomplish this by running interference with another sound. If your tinnitus sounds very abrasive, like fingernails on a chalkboard, we may be able to replace it with a pleasant sound like ocean waves or wind chimes. There are a number of these sound therapy strategies on the market, and your audiologist should be able to help you decide which device would be best for you and then tune the sound therapy to provide the maximum benefit.

If you have tinnitus and also need hearing aids, just wearing the hearing aids alone may be enough to interrupt tinnitus and provide relief. However, you also might want to consider getting a combination hearing aid and tinnitus sound therapy device, usually called a *tinnitus instrument*. These devices act as a hearing aid but *also* can be programmed to put out a sound therapy—to disrupt the tinnitus when you want it to. This device does not get rid of the tinnitus, but it may cover it up with a second sound that is more pleasant or can distract you from the tinnitus. This allows your brain to stop focusing on the tinnitus and to stop viewing it as a threat or a danger, which in turn allows you to push the tinnitus into your subconscious.

A tiny minority of tinnitus sufferers find their tinnitus so disturbing that it interrupts sleep, causes depression and anxiety, and in extremely rare cases even causes suicidal thought. It is interesting to note that Vincent van Gogh had a disease that caused chronic tinnitus, which was very disturbing to him. Some people postulate that he cut off his ear in an effort to make the tinnitus stop.

A number of therapies and medications can help with the mental health side of tinnitus. If you are having depression or thoughts of harming yourself, please bring this to the attention of your physician, your mental health professional, and your audiologist as soon as possible. Help is available.

Would learning to lip-read help me?

Lip-reading used to be a more common rehabilitation technique for people with severe or profound hearing loss than it is today. Unfortunately, lip-reading is not a particularly effective way of communicating. Only 20 to 30 percent of the speech sounds are visible on the face. Most of these sounds are produced back in the throat, where the viewer simply cannot see what's going on. Even for the speech sounds that are visible, many will look identical. For instance, *p*, *b*, and *m* all look exactly the same on the face.

I used to work in a clinic where we would take people through a 16-week course in lip-reading. They would meet multiple times a week and end with approximately 45 hours of training. We found that their ability to lip-read went up maybe 1 to 2 percent. This is not an effective therapy for most people.

How about sign language? Who can benefit from learning to use it?

Sign language can be an extremely effective mode of communication for patients who cannot communicate verbally even with the best hearing devices or surgeries, or who simply want to be a part of the Deaf culture and the Deaf community. Be aware, though, that learning sign language really is learning a whole new language. Sign language has its own grammar, vocabulary, and culture.

Most people who acquire hearing loss later in life are not terribly successful with sign language for the simple reason that no one else in their life speaks it. It usually is not helpful in connecting you with your existing community of oral communicators.

Children, however, often benefit greatly from learning sign language. Teaching sign language, either with or without oral communication skills, usually gives young hearing-impaired children faster access to language than hearing alone can.

If you have a young child who has a hearing loss, consult with a deaf educator, a speech language pathologist, and an audiologist about which communication mode may be best for your child. The best route may be oral education alone, sign language alone, or a combination of sign language and oral

communication, which is called *total communication*. The decision is a very individual one that should be up to the parents. The goal of this book is only to familiarize you with the concept of deaf education and to encourage you, particularly if you have a child younger than two with a hearing impairment, to seek out professionals in your area to get firsthand advice about what is best for your child.

Is there any benefit to active listening therapy?

An increasing number of computer-based therapy programs are being marketed to use what's called neuroplasticity, or the ability of the brain to restructure itself with experience. These therapies can be used to help improve memory, to help you with problem solving, or even help hold off dementia or Alzheimer's. We're also seeing neuroplasticity therapies that are said to help with listening tasks.

A number of computer-based programs use guided and focused listening tasks to try to optimize your neurologic ability to interpret sound. The idea is that most people put off getting hearing devices for years, allowing their brains to become less specialized for hearing sounds because of disuse. These programs are supposed to be like physical therapy for your brain; if you can retrain your brain to make better use of sounds, your speech understanding will go up.

This all sounds good on paper, and it may pan out in the real world. Currently, though, there is not, in my opinion, enough data to show that active listening therapies create a real world benefit. If your audiologist offers an active listening therapy that is low cost, by all means try it to see if you feel it's a benefit.

I would also suggest that if you are considering an active listening therapy that costs thousands of dollars, ask the prescribing professional for data that proves that the therapy is effective. You as the consumer should be comfortable in requesting solid information on your prognosis if you undergo this therapy.

I don't like my hearing aids. Is there any point in taking them to a new audiologist to see if something can be done to allow me to hear better?

As an audiologist, I spend a good deal of time working with people who have spent a good deal of money elsewhere on what could be considered failed hearing aid fittings. Many of these patients did not make an informed or smart choice about who they did business with. Instead of getting health care, they wound up buying a device that was improperly set for them.

You may have spent thousands of dollars on hearing aids that you cannot wear. You may have found that there's too much background noise, or that you sound like you're talking in a barrel, or that things just aren't clear. If this is the case, do not despair. Hearing aids frequently can be reprogrammed or readjusted by another provider to better meet your needs.

Hearing aids are much more than just a device. When you purchase hearing aids, what you're really purchasing is the provider's expertise on how to program and set these devices to best process sounds for your individual needs.

When your audiologist gets hearing aids from the manufacturer, they don't come preset for you. They can be programmed to fit a number of different levels of hearing loss—anywhere from mild all the way up to severe. If the professional that you're dealing with doesn't know how to program those hearing aids, you can wind up with wildly inappropriate sound being pumped out of them and into your ears.

If a professional ever tells you, "Just get used to it. You just need to wear them more," you're probably dealing with a professional who doesn't know how to adjust the hearing aids well. There should be a give and take in setting your hearing aids.

It's normal to expect three or four follow-up visits after the initial fitting with a hearing device to make it sound as good as possible for you. The professional will probably try to stretch your comfort zone and maybe push you to accept a little bit more sound than you're 100 percent comfortable with initially. The idea is that your brain and your ears will become accustomed to that sound.

But if you're seven or eight appointments in and you're still not satisfied, it's time to find someone else to help you with your hearing aids. If your hearing aids still sound terrible several months into wearing them, pick up the phone and call someone else. Often all it takes is a new set of eyes—maybe a new skill set—to look at how the hearing aids are programmed, listen to your particular concerns, and make a compromise so that the hearing aid can deliver more of what you're hoping for. You should never have hearing devices sitting in a drawer.

A lot of people assume that once they have a pair of hearing aids, another audiologist would not be willing to program or adjust those hearing aids for them. They worry that it would be rude or in bad form to switch to another health care provider.

Your audiologist is here for you. If you are not feeling that you're clicking well with an audiologist, or that your concerns are not being addressed well, it is up to you to find somebody who addresses your problems better and switch to that provider. As health care professionals, we know that none of us is going to be the perfect fit for everyone; we all have unique personalities and skill sets. I might refer a patient to another clinic across town if I feel I'm not the best fit, and other clinics will refer patients to me if I have something that I can offer that they cannot. This is not personal. This is about getting you to the healthcare that you need.

You do need to be prepared for the fact that asking an audiologist to spend hours of appointment time with you to program hearing aids that you purchased elsewhere will likely carry a cost. You should not expect an audiologist to spend hours adjusting your hearing aids and not receive a bill. You purchased the hearing aids from somebody who could not complete the job to your satisfaction and you paid them money, but the next audiologist along the way should still be compensated. Audiologists commonly have an hourly charge to reprogram hearing aids that were purchased elsewhere.

The reprogramming fee, though, is typically a fraction of the cost of buying new hearing aids. So if you have hearing aids that you are dissatisfied with and that are sitting in the drawer, you owe it to yourself to

see if there's anything that can be done so that they work for you. The cost to readjust the hearing aids will typically be well worth it in the end.

Chapter 7

How Do Hearing Aids Work?

Choosing the right type of hearing aid is incredibly important. Earlier this year a patient, Matthew, came to me who had been fit with the smallest hearing aids on the market. They were completely in the ear canal. The reason that he was fit with this kind of hearing aids is that he went to his hearing aid dealer and said he didn't want his hearing aids to be noticed. The hearing aid dealer was happy to accommodate him, because that particular office had its highest profit margin on tiny, completely-in-the-canal hearing aids.

Unfortunately, this type of hearing aid was a disastrous choice for Matthew. His particular hearing loss was completely inappropriate for this kind of hearing aid. As soon as he put the hearing aids in his ears, he felt like he was talking with his head in a barrel or with his ears plugged up with ear plugs. Needless to say, he immediately stopped wearing them. He had spent over \$7,000 on those hearing aids, and they were sitting in a drawer.

When he came to me, we first talked about what he thought he wanted. Then we talked about the different types of hearing aids he could have and what would actually meet his needs.

In the end, we selected an open-fit hearing aid, which is a type of behind-the-ear hearing aid. It leaves the ear canal wide open, allowing the wearer's voice to sound much more natural. It also improved Matthew's speech understanding and background noise in places like restaurants.

If Matthew had had the knowledge that you are getting by reading this book now, he would have been able to go in and effectively communicate his problems to his hearing aid dealer in the first place. The end result would have been less frustration and more success, and he would have saved a lot of money.

This chapter provides an overview of the main types of hearing aids. We also discuss many of the specific features that may make one particular hearing aid more effective than another and why you might choose an in-the-ear hearing aid versus a small behind-the-ear hearing aid. This information is detailed and changes rapidly as technology improves. You will certainly want to follow up with your audiologist regarding specific features when considering new hearing aids.

How does modern technology create better hearing aids and make my life better?

Hearing aids used to be simple amplifiers. They simply made every sound louder. They could not be tuned effectively to your particular type of hearing loss.

Modern hearing aids, though, are fit by prescription. They are fit to your exact hearing loss and target the types of sounds that you don't hear. Most commonly, those are soft, high-pitched sounds.

What you probably need in hearing aids is not simply amplification; you need amplification plus volume control. When you're missing soft, high-pitched sounds, you need amplification for the softer portions of speech (*s*, *f*, and *th*) but not for the louder portions of speech. When effectively fit, modern hearing devices take this into account. They meet your individual needs and selectively amplify just the sounds that you're missing. The end result is that with a modern hearing device, you should feel not so much that things are louder, but that things are clearer. This means you'll be more satisfied.

How does the audiologist know what I need?

When we fit modern hearing devices, we use what's called a prescription formula—basically a mathematical formula that tells us how much amplification you need at each different pitch for best speech understanding. We begin with the results of your hearing test, or audiogram (see Chapter 4). If you have normal hearing for deep, bass sounds, we don't give you any additional power for those normal hearing pitches. If you have a substantial hearing loss at a particular pitch, we try to give you amplification equal to approximately half of your hearing loss back at that pitch.

This may sound surprising. Hearing aids do not give you 100 percent of your hearing loss back. If you have a 50 dB hearing loss, we give you less than 50 dB of amplification. Most people think that hearing aids should bring your hearing back to fully normal hearing. If we gave you access to all the sounds that you've been missing, you would be overwhelmed. You would feel like you were inundated with background noise and that you couldn't hear through it at all. By bringing back approximately half your hearing loss in terms of amplification, we're able to present you with the most important speech sounds without overloading you with background noise.

It's important to remember that hearing aids don't get you back to normal, but they should get you a good deal closer than you were without them.

What choices do I have for shapes and styles of hearing aids?

One of the things that people think about most when they're considering hearing aids is cosmetics: How small can the hearing aid be? How invisible can it be?

Although cosmetics are important, it's equally important, or even more important, to get hearing aids that perform well—that help you understand speech. After all, that's why you are going to the audiologist.

There are three main styles of hearing aids to consider: behind-the-ear, in-the-ear, and completely-in-the-canal hearing aids.

A behind-the-ear hearing aid has its electronics outside and, obviously, behind the ear, and a small tube or wire goes down into the ear canal to direct sound to the ear drum. This kind of hearing aid can be very large if the hearing loss is severe to profound, or it can be so small it is almost invisible in people with mild hearing loss—and anywhere in between.

Today, approximately 85 percent of people prefer some type of behind-the-ear device (sometimes referred to as BTE, mBTE, RIC, RITE, or mRITE). Even though this style is more noticeable than the smallest completely-in-the-canal styles, it often sounds more natural. By moving the electronics out of the ear, we are able to open up a large tunnel through the earpiece in the ear canal. This allows your own voice to sound more natural to you. People with in-theear hearing aids sometimes say it sounds like they are talking into a barrel. This is not a problem with behind-the-ear devices.

In addition, behind-the-ear hearing aids are typically as comfortable, if not more comfortable, than the in-the-ear styles we'll discuss next. Because the bulk of the electronics are not in the ear canal, there is less physical mass and pressure inside the ear.

The second type of hearing aids—in-the-ear hearing aids, sometimes called custom made hearing aids—fill up most of the ear canal. These gained popularity in the 1970s and 1980s, when behind-the-ear hearing aids were extremely large, heavy, and uncomfortable. However, the in-the-ear style made wearers' voices sound quite distorted when they talked.

Then, in the 1980s, President Ronald Reagan came out of the closet and told everyone that he had a hearing loss and wore hearing aids. He wore a type of hearing aid called the "in-the-canal" hearing aid, or ITC hearing aid. Once Reagan let everyone know that even the president of the United States could have a hearing loss and be effective, everyone asked their audiologist for the president's hearing aid. ITC hearing aids became very popular.

CIC hearing aids are tiny—the closet thing we have to invisible hearing aids, even smaller than the ITC style that President Reagan wore. They generally sit entirely in the ear canal. If you look closely inside a wearer's ear, you will probably find the hearing aid, but they are very difficult to see. From a cosmetic standpoint, a CIC hearing aid is the best option.

These hearing aids, however, have some significant negatives. Because they block the ear canal, they will make your own voice sound very distorted to you; it will sound like you have a terrible head cold. To get an idea of what this would sound like, put your fingers in your ears as you're

reading this, and say a few words. You will immediately know what I'm talking about when I tell you that your own voice will sound distorted with a CIC hearing aid.

The other negative is that some of the newest hearing aid technologies are not available in the CIC devices, so you often lose some performance, particularly in noisy environments, by stressing cosmetics instead of technology. One of the most important technologies that are not included in CIC devices are directional microphones to improve hearing in background noise, which will be discussed later in this chapter. The reason they cannot be included in CIC devices is that the two microphones usually need to be at least 6 mm apart to help you focus more on what's in front of you and less on what's behind you. A completely-in-the-canal hearing aid does not have enough room to put those microphones in that configuration. So if you would like the best possible hearing in moderate amounts of noise, such as a restaurant, a group meeting, or a family gathering, you will be happier with a hearing aid larger than a CIC to achieve the best hearing in a background noise situation.

What features do today's hearing aids have to help me hear better?

Hearing aids today come with a wide range of possible features and a wide range of prices. In general, as you go up in price, you're purchasing better technology. This means that you will often get some advanced features in the higher priced hearing aids that you won't get in lower priced ones. This is important for you to know, because these features will make a big impact on your ability to understand speech in your most challenging listening environments.

The number of possible features can by dizzying. They include automatic volume control, digital noise reduction, directional microphones, features that will make your own voice sound more

natural, features to eliminate squealing sounds, binaural (two-ear) signal processing, frequencylowering techniques, remote control features, wireless screening options, and water and dust resistance.

Some features, such as automatic volume control, are found in almost every hearing aid on the market. Others, such as binaural signal processing, advanced noise suppression, and the most effective directional microphones will only be found in the most expensive hearing aids. But you may not need to get the most expensive hearing aid on the market—you may be able to economize without giving up performance in your most important listening environments.

To decide whether you need these features, talk with your audiologist about your specific listening requirements and about what you're having difficulty hearing. You audiologist then can determine which features would most effectively address your individual problems. Also, the audiologist should help determine whether there are features that you do not really need. More features are not always better. Sometimes getting all the different options can be a lot to manage when you just want the hearing aid to adjust to your needs automatically. If you tell your audiologist what you want to accomplish, they can select a product that gives you the RESULTS you need instead of the FEATURES that sound like a good idea.

What is automatic volume control?

Almost every hearing aid on the market has some sort of automatic volume control. The hearing aid can monitor how loud the incoming sound is and adjust itself based on that. The hearing aid will try to keep soft sounds audible, moderate conversation comfortable, and loud conversation tolerable. When a loud sound comes in, the hearing aid turns the volume down to keep you in your comfort zone. If the room is quiet, the hearing aid pushes the volume a little harder. You never have to touch the volume control; it is automatic. This type of approach will generally keep the hearing aid in your comfort zone 98 percent of the time.

Some people, though, would prefer to control the volume themselves. They prefer a manual volume control, which allows them to constantly make their own decisions about how loud or soft the hearing aid should be. If you are nervous about having the hearing aid make volume adjustments for you, your audiologist can equip your hearing aid with a small knob or switch to allow you to control the volume yourself. That being said, most people are much more comfortable having the hearing aid make volume adjustments for them.

Many people find the best compromise to be a hybrid device. A hybrid hearing aid has automatic volume control but also gives you the ability to override what the hearing aid thinks you need in situations where you want a little bit more or a little bit less volume.

Is it worth it to get a hearing aid with a directional microphone?

The biggest problem that people have with their hearing loss is hearing while there is background noise. Directional microphones are, by far, the most effective way to deal with that problem. In a noisy environment, directional microphones go into a forward focus. That means they give you your most efficient amplification for what is within 8 feet in front of you. In a noisy restaurant, people at your table are generally going to be in front of you and within about 8 feet. There also will be a significant amount of noise coming from behind you. In this kind of situation, a hearing aid with a directional microphone will help you focus on the people you want to hear. It is very important that you understand that no hearing aid can ever simply amplify only the things that you want to hear. We don't have a hearing aid that can read your mind. What people typically want to hear with their hearing aids is speech. What they typically want to eliminate also is speech, only this time speech is labeled as background noise because it is a conversation we are not interested in at that moment. A hearing aid in a group of people, though, doesn't know who is interesting and who is boring—or who you would like to hear over the others.

If you're at a party and you want to hear Jim's story, Jim's voice is labeled as speech in your mind. The guy next to him, Joe, is labeled as background noise to you because you don't care about what Joe is saying at that moment. But five minutes later, when you're tired of listening to Jim and want to hear what Joe has to say, now Joe's voice, which used to be background noise, is categorized as speech in your mind. And Jim's voice, which used to be speech, is now labeled background noise. The voiced did not change at all. The only thing that changed was who you wanted to listen to. The hearing aid doesn't know when you change your mind about who you are interested, and cannot eliminate the voices just because you consider them boring.

Directional microphones can help solve this problem. With directional microphones, you get your most effective hearing for the person you are facing. In a party environment or a restaurant, you will typically face the person you want to hear, naturally orienting the hearing aid toward the speech you think is most important. So while the hearing aid can't read your mind, it can to an extent read your body language and help you most efficiently amplify the person you are facing.

How does digital noise reduction help?

In the 1990s, the early digital hearing aids tried to amplify only sounds that were behaving like speech and to reduce the amplification for sounds that were not behaving like speech. This feature is called digital noise reduction. Basically, the hearing aids analyzed the number of peaks in volume of the incoming sound. For example, the device would recognize that in running conversation, there are three or four peaks per second in volume. If the incoming sound has a lot more peaks in volume or a lot fewer peaks in volume, the hearing aid would decide that sound is not speech, and it would turn the volume down for those sounds.

Modern hearing devices do much more sophisticated analyses of the incoming speech sounds than the early devices. They are much more accurate at determining which sounds are acting like speech and which are not. The concept is still the same, though. If a sound behaves like speech, the hearing aid tries to leave the volume turned up; if a sound does not behave like speech, the hearing aid tries to turn the volume down for that particular sound in order to prioritize the important speech signal. Although digital noise reduction helps make noisy situations more tolerable, most of the speech understanding benefit in noise comes from the use of directional microphones.

Can I get a feature that will make my own voice sound more natural? Can the annoying sound of my chewing be lessened?

Some of the traditional and common complaints about hearing aids have been "When I'm wearing my hearing aids, my own voice sounds like my head is in a barrel" or "I sound like I'm

talking with a bad head cold." Many people say, "My chewing sounds abnormally loud to me when I wear my hearing devices, and it's annoying."

These problems are primarily the result of the hearing aids blocking the ear canals. As you'll recall from our discussion of types of hearing aids, most of this problem can be alleviated with a behind-the-ear hearing aid with its large tunnel or vent. Some hearing aids leave the ear canal so unobstructed that we call them open-fit hearing aids.

What is feedback control, and why might I want it?

Feedback is the squealing that occurs when amplified sound gets re-amplified over and over again. You'll often hear it on public address systems if they're turned up too loud, and a hearing aid is just a miniature public address system.

With hearing aid technology of the past, if we turned up a hearing aid too loud, some of the sound would leak out of the ear canal and get amplified by the hearing aid over and over again. By opening up a large tunnel through the ear canal, as we do with behind-the-ear hearing aids, we open up a path for that sound to leak out of the ear.

Modern hearing devices are able to combat the problem of squealing by incorporating feedback control circuitry that can detect when a hearing aid starts to squeal, at which time it creates an equal but opposite sound, canceling out the squeal. This technology allows us to get more effective power out of the hearing aid without its squealing. A by-product of that is that we are able to open up a large tunnel through the ear canal portion of a behind-the-ear hearing aid without subjecting you to squealing, so that your own voice will sound more natural to you. The development of feedback control technology has essentially made it possible to dramatically increase your ability to hear high-pitched sounds while still making your own voice sound more natural.

Is a binaural, or two-ear, signal processing feature useful in a hearing aid?

One of the biggest advancements in hearing aid technology in the last few years is the ability of the hearing aids in each of your ears to act as a team to help with noise reduction and increase the clarity of speech in noisy environments. This is called binaural signal processing; it essentially means that your two hearing aids communicate with each other dozens of times a second. The right hearing aid talks to the left hearing aid, and the left hearing aid talks to the right hearing aid. If there is a dominant, loud voice on your right side, the hearing aids lower the volume on your left side, allowing you to more effectively focus on the dominant voice.

This feature can be very, very useful if you are trying to listen to one voice in a restaurant or party setting where there is, say, loud chatter on your left side and a dominant voice on your right side. The hearing aids can help coordinate the amplification on the left and the right to help you focus on one ear. This is, in effect, what your ears could do before they had hearing loss. In a noisy environment, your ears could "adjust" their internal volumes, allowing you to focus on one side or the other. Modern hearing aids with binaural signal processing can approximate that effect, although not as well as your natural biology could before you had hearing loss.

Could I be helped by frequency-lowering techniques?

In some kinds of hearing losses, people have frequencies or pitches at which the hearing loss is so severe that no matter how loud we make sounds, they will not be clear. Essentially, there's a blind spot in the hearing. This effect typically occurs at high pitches and often is caused by years of exposure to loud noise. In people with blind spots in their hearing, no matter how loud a hearing device makes high-pitched sounds, such as *s*, *f*, and *th*, the ear and brain will not hear them.

For people with blind spots in their hearing, hearing aid technology can take those high pitches and shift them to lower pitches at which the person's hearing is better. Patients often say that initially, the lowered speech sounds sound a little distorted, lispy, or slushy but that usually after four to six weeks, those sounds start to become more natural. The brain starts to accommodate the sounds and use them to be able to understand speech more effectively than it could with traditional amplification.

Do you recommend remote controls for hearing aids?

Some hearing aids have various settings the user can control. For example, you may prefer to get a hearing aid that allows you to switch on a music appreciation mode or a background noise reduction mode yourself rather than have your hearing aid do it automatically. Some users like to be able to manually control the volume of their hearing aids by pressing + or -. Most hearing devices are available with all of the controls on board the hearing aid itself, the controls in a remote control device, or both.

If you have a hearing aid that allows you to change settings, remote controls give you easy access to those different settings in your hearing aid. I have to admit that I have a love-hate relationship with remote controls for hearing aids. They can be incredibly useful for some patients and just an expensive gimmick for others.

Before you get a hearing aid that you adjust using a remote device, think about this: having all of the controls and adjustments available on the hearing aids themselves is incredibly convenient. If you're wearing the hearing aid, you have full access to all of its features at any time. You don't have to remember to bring your remote control with you when you go out or even go to another room in your house; you still have the ability to change the volume or to make adjustments to your hearing aids.

If you do not have good manual dexterity—for example, arthritis or numbness in your fingers or have vision problems, a remote control device can be a very useful option. A remote control is easier to manipulate. You do have to carry it with you, but if you cannot make the adjustments you'd like on your hearing aids due to dexterity or vision problems, a remote can be very helpful.

My experience has been that most patients absolutely love the remote control for three weeks. Then they get tired of carrying it around with them, and it ends up sitting on the nightstand at home. This means the hearing aid wearer does not have the ability to adjust the hearing aids except in bed, which is not very useful.

If you're getting a hearing aid with a remote control, then, I'd advise you to make sure the hearing aid also has onboard adjustments as well. That way, if you forget your remote control at home or decide not to carry it with you, you still have the ability to control the hearing aid by using the buttons or controls mounted directly on it.

Why might wireless streaming options be useful?

Modern hearing aids can now be set up to act as wireless headphones for a number of different sound sources. You might have sound sent directly from your cell phone, TV, computer, iPad, MP3 player, or stereo to your hearing aids, in essence turning them into a set of earphones that make the sound much clearer. This is similar to using a Bluetooth earpiece with your cell phone. The difference is that the audio signal can be sent to both ears and amplified by the hearing aid to dramatically improve the clarity.

Almost every major hearing aid manufacturer currently has the option of a wireless streaming device for use with your hearing aids. The wonderful thing is that you, the hearing aid wearer, can then have individual control over how loud you want to hear an electronic sound source. If you're watching television with your spouse and want to turn up the volume, you can do so using the wireless streaming option for your hearing aid, and your spouse can listen at the volume desired. This option, obviously, can eliminate a lot of arguments in the home over the desired TV volume.

Would any features be able to help with my tinnitus?

Many people with and without hearing loss are bothered by tinnitus, a sound that you can hear but that is not actually in the world around you. Buzzing, roaring, ringing, hissing, or thumping noises are commonly reported.

As we said in Chapter 6, if you have tinnitus and also need hearing aids, just wearing the hearing aids may be enough to interrupt tinnitus and provide relief, but tinnitus instruments (devices that combine hearing aid and tinnitus sound therapy) might be helpful. These devices act as a hearing
aid but also can be programmed to deliver an interfering noise through the earpiece—usually static or a semi-musical sound—to cover up the tinnitus or distract you from it when you want it to. Your brain is able to stop focusing on the tinnitus and stop viewing it as a threat or a danger, which in turn allows you to push the tinnitus into your subconscious.

Should I look for a hearing aid with water and dust resistance?

Today's hearing aids are much more resistant to water, dirt, and dust than in the past. Most hearing aids now are designed to be water resistant, which means they can tolerate a brief submersion in one meter of water. In practical terms, that means if you got your hearing aid wet or even dunked it under water as deep as three feet, it would probably come back to life if you opened up the battery compartment and let it dry out for a day or two.

You might be wondering why depth of water is more important than how long a hearing aid is underwater. The deeper the water, the more force the water pressure exerts on the hearing aid, and the more likely it is that water will infiltrate the hearing aid.

Since most hearing aids are water resistant, they will stand up to brief exposure to water or rain. For example, if you get caught in the rain or forget to remove them before showering, they should be fine as long as you open them up, take out the battery, and let them dry out. Try not to put the hearing aid to the test, but they are designed to withstand brief exposures to water.

Some people would like even more water resistance. If you participate in activities that run the risk of getting submerged in water—fishing, white water rafting, kayaking—you may want to

order waterproof hearing aids. These hearing aids continue to run even after submersion in water and are rated to even greater depths of submersion. Some are rated as submersible in three to nine meters of water and will still function when you take them out to dry.

If you choose waterproof hearing aids, however, be aware that you probably will have to make some sacrifices in sound quality. You may have to trade off some of the sound processing strategies we normally use for improved clarity in order to get a waterproof hearing aid.

What Can I Expect from My Hearing Aids?

- Cosmetics you feel good about
- Improved speech understanding, even in moderately noisy environments like the car and in restaurants
- Easier communication at a distance, including across the room, in a classroom, or in a lecture setting. Little to no whistling, squealing, or feedback
- Comfort that allows you to wear them most waking hours of the day without thinking much about them
- Improved quality of life
- Soft speech should be audible, conversational speech should be comfortable, loud speech should be tolerable

How can I choose one brand or manufacturer over all the others?

At this time, approximately seven manufacturers make the vast majority of hearing aids sold in the United States. The truth is, every single one of these manufacturers makes a good quality product. What they differ in is their ideas about what someone with a hearing loss needs in terms of sound processing.

The truth is that no one manufacturer makes the right hearing aid for everyone, but each one makes the right hearing aid for someone. Therefore, when choosing a hearing aid, be sure that you go to an audiologist who carries several manufacturers' brands. These audiologists will be able to advise you, based on your hearing test, which manufacturer is most likely to fit your needs; they will try to find the right hearing aid for you rather than trying to fit your hearing loss into what one manufacturer has to offer. Chapter 2 gives other tips on finding a great audiologist for your needs.

Chapter 8

What Insurance Coverage and Legal Protections Will I Have When I Purchase Hearing Aids?

In spite of my office's best efforts to make sure that everyone understands their protections in buying hearing aids, almost every year a situation comes up that could have been avoided. Earlier this year, a patient, who I'll call Sarah, came in and told me that she had purchased hearing aids about five years ago. Sadly, about three months into wearing the hearing aids, her dog ate them—crushed them beyond recognition. She was devastated. The hearing aids had been a huge help. They had improved the quality of her life dramatically. But they were expensive. She knew she couldn't afford a second pair of hearing aids and just suffered in silence as she saved her money for the next set. Now, five years later, Sarah finally had enough money and came in to see me to get her next set of hearing devices. If she had only known that almost every hearing device sold in the United States comes with loss and damage insurance! This means that in that first year of ownership, she could have gone back to her hearing aid provider and got a brand new replacement set of hearing aids for little or no cost.

Every hearing aid sold in the United States will come with a printed list of all of your rights your right to free repairs, your right to replacement in the event of loss or a dog eating the hearing aids, and your right to return the hearing aids for a refund. Be sure you read this information when you get your hearing aids. You will be presented with an enormous amount of information when you first get your hearing device, but your rights are among the most important part of this information.

This chapter outlines your rights when you buy a hearing device. A number of legal protections ensure that your hearing aids are successful for you and that you are successful with them. The bottom line is that if there's ever a problem with your hearing devices—even if you lose them, you run them over with a car, or the dog eats them—get in touch with your audiologist and find out what protection you have.

Hearing aids are a major expense. Your insurance coverage may cover a portion of the purchase price. In addition, the hearing aid industry is regulated by the government, and a number of laws are out there to protect you, the consumer, as you purchase your hearing aids. Understanding your legal protections will help you get the hearing aids that are best for you.

How can I know if my insurance will cover anything related to my hearing aids?

Almost 30 percent of the new patients in my office have some form of hearing aid coverage through their health insurance. The insurance usually doesn't cover the entire cost of hearing aids, but it often covers \$1,000 to \$4,000, which is certainly a big chunk of the hearing aid purchase.

When you call in to consider an appointment with an audiologist's office, be sure to tell them what health insurance you have. You will want to find out if that office is what's called a preferred provider for your insurance. If they are a preferred provider, it means that they are contracted with your insurance company to give you the best possible price to maximize your insurance benefits. This may mean that the co-pays that you pay are lower or that your deductible is lower. In some cases, in fact, your insurance may *only* provide a benefit if you go to a preferred provider. You may find out that your insurance company will not pay for you to go to the first office you call. Therefore, it is important to discuss your insurance before you make an appointment to go in for a hearing test.

Be sure to ask the audiologist's office up front whether your insurance will cover the cost of your exam, and ask them to check and see if the hearing devices would be covered as well. That way, when you're figuring out what device you want to purchase, you will have a clear understanding of what your out-of-pocket costs will be.

I have Medicare. What will it cover in relation to my hearing aids?

Medicare typically covers the cost of a hearing test if it is medically necessary. It will not cover the cost of hearing aids, however. It also does not cover routine exams (exams that are not medically necessary). The definition of *medically necessary* is not clear cut.

In general, though, if you tell your primary care doctor that you are truly concerned about your hearing and you would like a hearing test to find out whether medical treatment is available, you should get a referral for a hearing test that Medicare will cover.

Medicare will not cover hearing tests solely for the purpose of getting a hearing aid. This means that if you know that you have a hearing loss already, and you know that it is not medically treatable but instead you want a hearing aid, Medicare will probably not cover your hearing exam. However, if you suspect you have a hearing loss and want to find out what all your treatment options are, Medicare typically will cover the cost of the exam.

The preceding discussion assumes that Medicare is your primary insurance. If you have a Medicare Replacement Plan (often called a Med Advantage plan) instead of Medicare—for example, Blue Cross Blue Shield's Medicare Advantage or Providence Health's Medicare Replacement Plan—you might have a hearing aid benefit or partial coverage for a hearing aid purchase.

What to ask your insurance carrier before you get hearing aids or, better yet, before you sign up or renew your plan:

1.) Is hearing health care covered under this plan? (If not, ask if they have a similar one that is covered.)

2.) Are there any age restrictions on who can exercise this hearing aid benefit? (It will do you no good if they only cover hearing aids for children and you're 45.)

3.) What kind of deductible will I need to meet in order to have you pay out on a hearing aid claim?

- 4.) What is the allowable maximum for hearing care under this plan?
- 5.) What kind of co-pay will I be expected to pay?
- 6.) Can I see any provider or does it have to be one in your network?
- 7.) Does this plan cover repairs, batteries, or replacement?

These are all great questions to ask your insurance company before signing up or before renewing your plan.

What if I decide not to get hearing aids after I take a hearing test? What will I have to pay?

Any reputable audiologist will charge for a hearing test. The office should be willing to tell you the cost for the exam prior to your appointment.

Hearing testing is not terribly expensive. It's probably the cheapest part of the process of getting hearing devices. Do not decide whether to go to a particular provider based on how much the initial hearing test will cost. Newspaper ads may tout free hearing tests, but ask yourself, "How is this person making a living?" No one can make a living giving free hearing tests 100 percent of their time. If you are considering going to somebody who is advertising a free hearing test,

keep in mind that the only way that they can make their paycheck that month is if they convince you to buy a hearing aid, whether you need it or not.

When you go to an audiologist to talk about hearing aids, you should be prepared to actually pay for that time with the provider. In the end, being willing to pay a small amount for a hearing test will better ensure that you will get honest, unbiased opinions about whether it's time to get a hearing aid.

What if I don't like my hearing aids after I get them?

Patients sometimes come into my office with the same complaint: "I bought a set of hearing aids but they don't work for me, so I just put them in the drawer."

This should never happen. You should never, ever be one of these patients who buys a hearing aid and winds up putting it in a drawer because you do not like it. A number of laws protect you, the consumer, when you buy hearing aids. This following section on legal protections discusses some of the most important protections that you have in buying a hearing aid so that they stay in your ear and not in the drawer.

What legal protections do I have when I purchase hearing aids?

The US hearing aid industry is regulated by a number of laws. When you purchase new hearing aids, you must receive a trial period and will almost certainly receive a warranty period and a loss and damage insurance period. Read on to learn about your legal rights in each of these periods. If you are aware of your rights, you will be protected from purchasing hearing aids that do not work for you.

How does the trial period for my hearing aids protect me?

Every hearing aid sold in the United States comes with a trial period. This is a period of time during which you can try out the hearing aid in your real-world situation—in your daily life—and decide whether this hearing aid is making your quality of life better. You can learn how your hearing aid sounds at home, at work, and anywhere else you go. In short, you have this time to decide whether or not you would like to keep the hearing device or return it for a refund. Federal law specifies that you must have a minimum of 30 days to try out your hearing aid. It's not uncommon to find a 60-day trial period, and occasionally you'll find an office that will give

a 90-day trial period. During that period of time, if you decide that you don't want to keep the

hearing aid for any reason, you may return it for a refund.

If you do return the hearing aid for a refund during your trial period, you should expect a few expenses that may not be refundable. Typically, the cost of the hearing test is not refundable, although in some states if you return a hearing aid the hearing test charge may also be refunded. Most states allow the audiologist to retain a small amount of the purchase price as a restocking fee to cover things like the audiologist's time and shipping costs. This is always specified in state law. Most states put a cap on the amount of money that the audiologist can hold back in the event of a hearing aid return. Some states cap the restocking fee by a dollar amount, such as \$100 per hearing aid; other states will specify the restocking fee in a percentage, as in up to 10 percent of the purchase price of the hearing aid.

Your purchase agreement should specify any restocking fee or other nonrefundable fees. Your audiologist will give you these details for that specific office.

What does my hearing aid's warranty mean to me?

Hearing aids typically come with a one-, two-, or three-year warranty. During this time, the manufacturer must fix any problems that the hearing aid has. These repairs should cover things like earwax working its way into the hearing aid or accidental moisture damage. It should cover repairs if your hearing aid just doesn't sound right, the hearing aid works intermittently or sounds distorted, or the device stops working for any reason. These types of repairs are considered normal wear and tear, and your warranty should cover them in full during that first one to three years.

You can purchase extended warranties for hearing aids. Most manufacturers will extend the hearing aid warranty out to as long as five years. You can expect to spend a few hundred dollars per year to extend the warranty on a pair of high-tech hearing aids.

I am not a big fan of extended warranties, because hearing aids are fairly durable. If you take reasonable care of them, you'll generally spend less on hearing aid repairs throughout the lifetime of the device than you would on extended warranties. If I were in your shoes, I would simply put a couple hundred dollars a year into a savings account, and pay for your repairs out of that money. At the end of the lifespan of your hearing aid, perhaps after five years, you'll usually be several hundred dollars ahead.

How does the loss and damage insurance period for my hearing aids help me?

Your hearing aids, by law, come with a loss and damage insurance period. This is separate from your hearing aid warranty and is designed to cover severe abuse and neglect. Loss and damage

insurance is designed to provide you with some protection if you lose the hearing aid or it sustains catastrophic damage. You will be covered if your hearing aid is lost, burned up in a fire, or fed to the dog. You could eat it, beat it to death with a sledgehammer, or drop it down the garbage disposal. You would be surprised at what people do to their hearing aids.

So even if you have obviously abused or neglected your hearing aid, and even if the hearing aid comes back to the audiologist in 15 pieces, your loss and damage insurance will provide a new hearing aid during the loss and damage period. This period is typically one to two years, and this kind of insurance typically comes standard in a new hearing aid purchase. The loss and damage insurance typically covers one replacement on each hearing aid during the insurance period. If you lose your right hearing aid and have it replaced, you still have coverage on your left hearing aid during this period.

Most loss and damage insurance policies have a deductible to pay the claim. It may be a dollar amount, such as \$150 per hearing aid, or it may be a percentage, such as 10 percent of the purchase price of the hearing aid. But certainly, in the event of a catastrophic event, a deductible is much, much less expensive than buying a whole new hearing aid.

The bottom line is that if you lose or crush your hearing aid, don't panic. Call your audiologist. Find out if you still have loss and damage insurance, and see if you're eligible for a replacement. There's a limited time to do this, and the clock is ticking, so don't delay. Nothing is worse than having somebody come into my office and tell me they lost their new hearing aid two years ago and that now they're ready to buy a new one. They simply could have had a replacement provided at little to no cost had they acted right away.

Loss and damage insurance can be extended beyond the first few years through the original hearing aid manufacturer or a third-party insurance company. Again, I'm not a huge fan of

buying extended loss and damage policies. Extended loss and damage insurance tends to be fairly expensive. If you're a reasonably responsible adult, you will probably keep good track of your hearing aids, especially if you're a full time hearing aid user. However, if you are worried about losing or damaging your hearing aids and the expense involved with replacing them, you might want to consider purchasing extended loss and damage insurance. For example, if you're blind, if the hearing aids belong to a child, if you're in assisted living, or if you have memory problems or tend to lose your glasses and keys, extended loss and damage insurance may be perfect for you.

An alternative to loss and damage insurance that many people don't think about is itemizing your hearing aids on your homeowners or renters insurance policy. Homeowners insurance or renters insurance tends to have a much cheaper yearly premium to insure your hearing aids, but you will probably face a much higher deductible in the event of a loss.

Do I have any protection if my hearing aids suddenly stop working well after the trial or warranty period?

It is common for hearing aids to suddenly stop working well at some point in their lifespan. They may have been perfectly satisfactory for two or three years, but then you notice that the device sounds weak, dead, or just odd. This situation can be due to a few different things.

Your hearing may have changed. After all, hearing changes all the time. Hearing loss typically is a slowly progressing condition, but it does tend to get worse over time. If you have a pair of hearing aids that seem to be gradually getting worse and worse, your hearing loss may have outgrown the current settings in the hearing aids. You should be getting your hearing tested every one to two years, at a minimum. If your hearing loss has worsened, usually your hearing aid can be reprogrammed or readjusted to match those changes in your hearing loss. Talk with your audiologist and set up an appropriate hearing testing schedule. If you visit your audiologist only when there's a problem, you can easily lose five or six years of improved quality of life by not addressing the changes in your hearing loss.

The other common reason hearing aids stop working well is mechanical problems. Most of these problems can be addressed very, very quickly. The most common problem that occurs suddenly is earwax that gets embedded into the part of the hearing aid that allows sound to get to your eardrum. The sound no longer can get into your ear. This problem is usually very, very easy to fix. If you have a hearing aid that was sounding great yesterday and today it sounds dead, your audiologist usually will be able to bring the hearing aid back to life in five minutes or less.

Sometimes a hearing aid has become so impacted with earwax that it has to be sent to the manufacturer for some new components. Even that, however, is a relatively easy procedure and much cheaper than getting a brand new hearing aid.

In general, if your hearing aid is five years old or less, you can expect it to be easily repairable and easily adjustable for your hearing loss changes. Never sit at home frustrated. If you have a problem, pick up the phone and call your audiologist. You're not being a nuisance. You are simply letting your audiologist know that you need some help. Your job as a hearing aid wearer is to be proactive. If you have a problem with your hearing aid, talk about it with your provider. It probably will help, and it can't hurt.

Chapter 9

Are Hearing Aids Really Worth All the Bother? Will They Really Help Me?

You may be thinking, "Okay, I understand a lot more about hearing aids now. I'm still not sure it's worth all the bother, though." To answer these concerns, I can only emphasize that professionally fitted hearing aids can make a dramatic improvement on your quality of life.

But do hearing aids really work that well?

I hear time and time again from patients, "I have a friend who has a hearing loss. He got hearing aids and didn't like them. Do hearing aids really work?" The short answer is in the vast majority of cases, yes, hearing aids really do work.

Approximately 80 percent of people with hearing aids are either satisfied or very satisfied with their hearing aids. The satisfaction rate reaches 96 percent with some hearing aids when properly fitted. Therefore, these hearing devices must be doing something positive. There is no way on earth that hearing devices can have an 80 percent or better satisfaction rating unless they're actually making a difference in people's lives.

How can hearing aids improve my quality of life?

You may remember our earlier discussion about the National Council on Aging study that found that almost every quality of life measure can be improved through the use of hearing aids in people with a hearing loss. Mental health, physical health, earning potential, relationships, and work performance can all dramatically improve through hearing aids.

Equally important is the fact that hearing aids can increase your overall safety and independence. For example, older patients often feel they must visit the doctor with someone along to make sure they hear and understand the medical recommendations. If you're not able to manage your own health care with your physician and must rely on others to act as your gobetween or interpreter, you're actually increasing the danger of misunderstanding your medication schedule, treatment recommendations, or surgery recommendations. You owe it to yourself to be as independent as possible in regard to your health care. You owe it to yourself to make sure that you can manage your medical appointments and be able to understand the decisions that you and your doctor have made.

Another benefit of hearing aids is increased confidence and ability at work. I cringe when people say, "I don't want to get hearing aids because I'm afraid that I will look less capable at work." Nothing could be further from the truth. Hearing loss is always more noticeable than a hearing aid. If you're responding inappropriately, or if you're smiling and nodding and faking it, people know what's going on. I have had so many patients tell me that after getting their hearing aids, "I no longer have to be afraid in group meetings. I no longer have to worry about whether I'm going to hear my boss or hear the board of directors. I know that I'm going to function well in work meetings and understand what's expected of me. As a result, I'm viewed as more capable at work."

If you have an untreated hearing loss, not only does your work performance and self-esteem suffer, but your income can also suffer. According to the U.S. Census Bureau, untreated hearing loss results in up to \$12,000 a year in lost income.

At this point, if you're still reading this book, you already know that you or someone that you love is struggling with hearing loss, and you're just trying to figure out if it's time to do something about it. It is indeed time to pick up the phone and call your audiologist, get a hearing test, and possibly even listen to what hearing devices can do for you. There's no reason to be yet another person who puts this off for years and years as your brain becomes less and less able to handle sound. You've put it off long enough.

You owe it to yourself to see if that is something that may be holding you back. Hearing aids are not just about hearing better. They change how people walk through the world and go through their lives. Hearing aids improve relationships. They improve work function. They allow people to stay in touch with their children and grandchildren.

Hearing aids are about more than hearing better. They are about making you into the person you can be and want to be.

GLOSARY

Air Conduction: presentation of sound by moving air molecules. This is accomplished with earphones or loud speakers. We normally hear through air conduction. This type of testing requires sound to travel through all three parts of your ear, the outer, middle and inner ear, before you can hear it. Any problems in any of these three areas will affect your air conduction hearing.

Amplification: in our context it is the act of making something louder. Amplification is sometimes used as another work for hearing aids.

Assistive Listening Device (ALD): any non-hearing aid device designed to help you hear or detect a sound. This could include a battery powered personal amplifier, a telephone amplifier, a flashing light to alert you to a doorbell or fire alarm, a vibrator to shake your pillow to wake you up if you cannot hear the alarm clock, etc.

Audiogram: a graph which indicates the softest sound you can hear for several different pitches. The audiogram is typically performed without hearing aids in to determine whether you have a hearing loss. Specific patterns of results on the audiogram can also help your audiologist understand what part of your ear may be affected by your hearing loss.

Audiologist: a university trained professional with specific expertise in hearing, hearing loss, hearing aids, and vestibular and balance testing. Any new audiologist in the United States must now have a doctorate degree to begin practice. That is usually a Doctor of Audiology (AuD) degree, but you may occasionally see an audiologist with a PhD or ScD degree as well. In the past audiologists were able to practice with a master's degree instead. You may see some audiologists still practicing with an MA or MS degree. That simply means they have been practicing since before the more stringent requirements for a doctoral degree were enacted for audiologists.

Bone Anchored Hearing Aid: a hearing device that is installed by a surgeon. It is usually used to treat either a conductive hearing loss, or hearing loss where one ear is essentially deaf, but the other is near normal. This device is not typically appropriate for the most typical type of hearing loss, sensorineural hearing loss on both sides.

Bone Conduction: testing that is performed by vibrating the bones of the skull instead of vibrating air molecules. Instead of testing with earphones or loud speakers, a small vibrator box is placed behind the ear or on the forehead. The vibrations travel through the skull to the inner

ear. This allows the audiologist to determine whether the hearing loss is due to a problem in the inner ear, vs. a problem in the middle/outer ear area. This information is vital to determine whether surgery may be a viable treatment.

Cochlea: the inner ear. This is the portion of the ear that converts sound from mechanical movement into an electrical signal that can be sent up to the auditory nerve to the brain. This is the part of the hearing system that is most often damaged with sensorineural hearing loss. In addition your vestibular function, which is vital to your ability to balance and stay upright is located in the cochlea.

Cochlear Implant: a surgically implanted hearing device used for very severe hearing loss. This device stimulates the auditory nerve with electricity instead of sound. This device is generally used when the hearing loss is so severe, that hearing aids cannot provide benefit.

Conductive Hearing Loss: a hearing loss located in the outer or middle ears. This type of hearing loss is a problem where the mechanical vibration of sound is damped down. In this type of hearing loss, the inner ear, or cochlea is still functioning well.

Decibel (dB): a measure of sound intensity. For most people higher numbers mean louder sounds. For instance a whisper may only be 20-30 decibels, but a gunshot may be 130-140 decibels.

ENT: ear, nose and throat physician. Some other names you may see for this highly trained professional are otolaryngologist (ear, nose, throat), otologist (ears only), or neurotologist (ears and brain). All of these refer to medical doctors who are trained to treat hearing loss. Their training specifically includes medications and surgeries, but carries only a little training on hearing aids themselves.

Hearing Aid: a device worn on or in the ear that is specifically programmed or tuned for your individual hearing loss. This is not a "one-size fits all" sound, but is fit by prescription to meet your individual needs. This type of device is designed for hearing impaired people with the intention of treating their hearing loss.

Hearing Aid Dealer: sometimes referred to as a hearing instrument specialist, hearing instrument dealer, or occasionally an audioprothetologist. In most states, these are retail sales people. Some states require some university level training, others do not. In general, the most rigorous training of hearing aid dealers result in BC-HIS after their name. Even with that designation, an audiologist will usually have six additional years of university training beyond the BC-HIS requirements.

Lipreading: the act of trying to understand speech by watching the shape of the lips and mouth. Lipreading is a very difficult task, and one that generally provides very poor levels of understanding.

Otolaryngologist: ear, nose and throat physician. See ENT section in the glossary.

Provider: a generic term that refers to any professional that is providing hearing testing or hearing aids. This could include audiologists, hearing aid dealers or ENTs.

Pure Tone: this is a very specific type of sound that is designed to emit just one frequency or pitch. We use this to stimulate a very specific part of the ear. Pure tones generally sound like beeps to the listener.

Sensorineural Hearing Loss: hearing loss located in the inner ear, auditory nerve or brain. This type of hearing loss is almost always permanent, and is typically treated with hearing aids or

cochlear implants. The most common type of sensorineural hearing loss is a loss of cells in the cochlea.

Speechreading: often confused with lipreading. This is the act of watching a combination of a person's lips, gestures, and facial expressions. This information can be used to enhance whatever portion of speech the person is able to hear.